

Remiern



Application of Artificial Intelligence for Surface Roughness Prediction of Additively Manufactured Components

Temesgen Batu ^{1,2}, Hirpa G. Lemu ^{3,*} and Hailu Shimels ⁴

- ¹ Department of Aerospace Engineering, Ethiopian Space Science and Geospatial Institute, Addis Ababa P.O. Box 33679, Ethiopia; temesgen.batu@kiot.edu.et
- ² Center of Armament and High Energy Materials, Institute of Research and Development, Ethiopian Defence University, Bishoftu P.O. Box 1041, Ethiopia
- ³ Department of Mechanical and Structural Engineering and Materials Science, University of Stavanger (UiS), 4036 Stavanger, Norway
- ⁴ Department of Mechanical Engineering, College of Engineering, Addis Ababa Science and Technology University, Addis Ababa P.O. Box 16417, Ethiopia; hailu.shimels@aastu.edu.et
- * Correspondence: hirpa.g.lemu@uis.no

Abstract: Additive manufacturing has gained significant popularity from a manufacturing perspective due to its potential for improving production efficiency. However, ensuring consistent product quality within predetermined equipment, cost, and time constraints remains a persistent challenge. Surface roughness, a crucial quality parameter, presents difficulties in meeting the required standards, posing significant challenges in industries such as automotive, aerospace, medical devices, energy, optics, and electronics manufacturing, where surface quality directly impacts performance and functionality. As a result, researchers have given great attention to improving the quality of manufactured parts, particularly by predicting surface roughness using different parameters related to the manufactured parts. Artificial intelligence (AI) is one of the methods used by researchers to predict the surface quality of additively fabricated parts. Numerous research studies have developed models utilizing AI methods, including recent deep learning and machine learning approaches, which are effective in cost reduction and saving time, and are emerging as a promising technique. This paper presents the recent advancements in machine learning and AI deep learning techniques employed by researchers. Additionally, the paper discusses the limitations, challenges, and future directions for applying AI in surface roughness prediction for additively manufactured components. Through this review paper, it becomes evident that integrating AI methodologies holds great potential to improve the productivity and competitiveness of the additive manufacturing process. This integration minimizes the need for re-processing machined components and ensures compliance with technical specifications. By leveraging AI, the industry can enhance efficiency and overcome the challenges associated with achieving consistent product quality in additive manufacturing.

Keywords: surface roughness; roughness prediction; additive manufacturing; machine learning

1. Introduction

Over recent decades, the manufacturing sector has gone through significant changes, yet it continues to play a critical role in the growth and development of both developed and developing countries [1]. It is considered the backbone of economies due to its contribution to economic growth, job creation, and technological advancement [2]. Additive manufacturing (AM) has gained acceptance among academics and the manufacturing sector as a powerful manufacturing tool. Recent research shows that it is more efficient than conventional production methods [3]. The manufacturing industry uses the term Additive Manufacturing to describe the process of fabricating physical objects from design data in a digital form by building them layer-by-layer. AM is a disruptive technology as it is not tied to individual production steps and does not require specific tools for each component, making it a universal production



Citation: Batu, T.; Lemu, H.G.; Shimels, H. Application of Artificial Intelligence for Surface Roughness Prediction of Additively Manufactured Components. *Materials* 2023, *16*, 6266. https://doi.org/10.3390/ ma16186266

Academic Editor: Silvio Genna

Received: 30 July 2023 Revised: 2 September 2023 Accepted: 7 September 2023 Published: 18 September 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). technology [4,5]. Numerous AM processes are currently available for commercial use. Various researchers have approached the classification of AM methods differently [6–8]. However, a widely accepted classification stems from the ASTM-F42 committee guidelines, which categorize AM into seven distinct groups [9]. These categories consist of vat photopolymerization (VP), material jetting (MJ), binder jetting (BJ), material extrusion (ME), sheet lamination (SL), powder bed fusion (PBF), and directed energy deposition (DED). The concise overview of all seven categories was presented in the paper [10].

The main advantage of additive manufacturing is that it can fabricate complex shapes and reduces material waste and production time (for some) [11–14]. Despite its advantages, additively manufactured products generally have poorer quality compared to those made with conventional manufacturing systems [15,16], primarily due to limitations in surface integrity [17]. Research indicates that in all additive manufacturing techniques, the improvement of surface roughness is a key objective. For example, vat photopolymerization (VP) and material jetting (MJ) yield parts with moderate surface roughness [10], whereas binder jetting (BJ), material extrusion (ME), sheet lamination (SL), powder bed fusion (PBF), and DED tend to result in parts with relatively poorer surface finishes [10]. Surface roughness is one of the factors encompassed by the concept of surface integrity. In Table 1, the surface roughness of most additively manufactured techniques is presented.

Table 1. Surface roughness of most additively manufactured techniques.

Name of AM Process	Minimum Layer Thickness		Surface Roug	ghness, Ra
	Values (mm)	Ref.	Values (µm)	Ref.
Stereolithography (SLA)	0.1	[18]	2–10	[19]
Selective Laser Sintering (SLS)	0.02	[20]	5-25	[21]
Fused Deposition Modeling (FDM)	0.05	[22]	0.5-20	[23]
Laminated Object Manufacturing (LOM)	0.10	[24]	0.8-2.5	[25]
Electron Beam Melting (EBM)	0.05	[26]	1-20	[27,28]
Direct Metal Laser Sintering (DMLS)	0.02	[29]	3–12	[30]
Binder Jetting	0.035	[31]	3–13	[32]
Direct Energy Deposition (DED)	0.25	[33]	5.08-227	[34,35]
Laser powder bed fusion (L-PBF)	0.02	[36]	3.5-13.45	[37]
Selective Laser Melting (SLM)	0.02	[38]	30–60	[39]

The surface quality of additively manufactured components is influenced by a multitude of factors that interact in intricate ways. These factors include the choice of AM technique, powder characteristics, layer thickness, scanning strategy, and energy parameters specific to each technique. For instance, in Vat Photopolymerization (VP), parameters such as layer thickness, exposure time, and resin properties contribute to surface quality [40,41]. Similarly, in Material Jetting (MJ), drop size, print speed, and layer thickness impact the surface finish [40,42,43]. In Binder Jetting (BJ), binder saturation and layer thickness play a crucial role in achieving desired surface characteristics [44–47]. For material extrusion-based methods in fuse deposition modelling or fused filament fabrication, parameters such as nozzle diameter, layer height, and extrusion speed influence the surface roughness [48–52]. The terms fused deposition modelling (FDM) and fused filament fabrication (FFF) are two terms that are often used interchangeably to describe the same 3D printing process. However, FDM is a trademarked term by Stratasys, while FFF is a generic term used by the open-source community [53].

Powder Bed Fusion (PBF) [54,55] and Directed Energy Deposition (DED) [56] are also characterized by their unique process parameters that have direct implications on surface quality. While the powder bed fusion techniques such as SLS, SLM, DMLS, and L-PBF (Table 1) are fundamentally identical processes that use a laser to selectively melt or sinter a bed of powder material to create a 3D object, there exist some differences in the way these processes work, such as the type of laser used, the power of the laser, and the scanning strategy used to melt or sinter the powder. These differences can affect the quality of the printed part, including its surface roughness. In general, thinner layers result in smoother surfaces, but can also increase printing time and cost. The optimal layer thickness for a given process depends on several factors, including the size of the powder particles, the power of the laser, and the scanning speed used to melt or sinter the powder. Table 2 presents the process parameters of selected additive manufacturing techniques.

Table 2. Process parameters influencing surface quality of AM manufactured components of most additively manufactured techniques.

AM Technique	Key Process Parameters Influencing Surface Quality	References
FDM/FFF	Layer thickness, Build Orientation, Raster Angle/Raster Orientation, Air Gap, Extrusion temperature, Print Speed, Infill pattern, Infill density/Interior infill, Nozzle diameter, Raster width, Number of contours, Contour width, Contour to contour Air gap	[51,57]
Powered-based fusion	Layer Thickness, Temperature, Laser Power, Beam Radius, Reflection Coefficient at Bed Powder Surface, Material Local Density, Velocity, Conductivity, Dynamic Viscosity	[55,58]
Direct energy deposition	Laser power, Scan speed, Powder feed rate, Travel speed, Layer height, Hatch spacing	[33,56,59]
Vat photopolymerization	Grayscale, Exposure time, Wavelength, Power source, Layer thickness	[60,61]
Binder jetting	Spreader speed, Layer height, Print head speed, Saturation level	[47,62]
PolyJet by Stratasys or MultiJet Printing (MJP) by 3D Systems	Print heads, Hardness range, Layer thickness, Material, Part size, Print resolution	[63]
Sheet lamination	Laser power, different exposure times	[64]

Furthermore, post-build treatments significantly contribute to the final surface quality of additively manufactured components [65,66]. For example, it can significantly improve the mechanical properties, i.e., surface roughness of the parts, making them more suitable for a wide range of applications [67]. Different post-processing techniques have been used by many researchers, such as shot peening (SP), laser shock peening (LSP), hot isostatic pressing (HIP), friction stir processing (FSP), and heat treatments (HT), to realize the better performance of the AM parts [66]. Various treatments, such as heat treatment, abrasive finishing, and chemical processes, can be applied to refine the surface texture and eliminate defects [68]. For instance, heat treatment can lead to grain growth and stress relief, affecting the overall surface roughness [63]. Abrasive finishing techniques, including sanding and polishing, can mitigate layer lines and irregularities, resulting in improved surface smoothness [68]. Chemical processes such as etching or electrochemical polishing can selectively remove material, enhancing surface finish [63]. Understanding the dependencies of surface roughness on these post-build treatments is crucial for achieving consistent and desirable surface qualities in additively manufactured components [69].

Surface roughness has been identified as one of the most significant factors affecting product quality, as stated in previous research [11]. For instance, Chan et al. [70] conducted a study to investigate the impact of surface roughness on product life and concluded that surface roughness leads to a reduction in product life expectancy. Moreover, surface roughness plays a role in the tribological behavior of surfaces [71], with rough surfaces experiencing faster wear compared to smooth surfaces. Therefore, it becomes crucial to predict and control the surface roughness of additively manufactured parts [64,72]. Additionally, surface roughness serves as an indicator for directly monitoring the mechanical characteristics of a workpiece, such as fatigue and surface friction, dimensional accuracy [73], and fracture resistance [74]. Understanding and managing surface roughness in manufacturing processes is essential for ensuring optimal product performance and longevity.

The impact of surface roughness on system performance is significant across multiple industries, such as aerospace [75,76], automotive [77,78], marine [79], semiconductor [80], chemical, wind energy [81], solar energy [82], food processing [83,84], and medicine [85,86].

Therefore, it is crucial to control the surface roughness of manufactured components. Thus, researchers give attention to predicting the surface roughness of additively manufactured components before printing/manufacturing the components. Predicting surface roughness in the manufacturing industry can lead to better quality control, cost savings, improved performance, and process improvement.

Different methods were used to predict the surface roughness of additively manufactured components, i.e., Taguchi-based regression models [87,88], statistical regression models [89], computational modeling (e.g., the FEM and Discrete Element Method (DEM)), [90] and machine learning methods [91]. There are several limitations of conventional (statistical) methods for predicting the surface roughness of additively manufactured components. Conventional methods for predicting the surface roughness of manufactured components often suffer from various limitations, such as a lack of flexibility, time-consuming nature, high cost, limited accuracy, limited applicability, inability to account for process parameter interactions, and limited understanding of underlying mechanisms [17,92].

However, the emergence of Artificial Intelligence (AI) approaches, including both machine learning and deep learning, has opened avenues to surmount these limitations. These AI approaches offer the potential to yield more precise and adaptable predictions of surface roughness. By harnessing these advancements in AI, particularly in machine learning, we can transcend the confines of traditional methods and attain the capability to design and fabricate high-quality components with unparalleled precision and efficiency. The flexibility, accuracy, and capacity to discern intricate patterns inherent in machine learning techniques empower us to overcome the challenges posed by conventional methods.

Machine learning methods have demonstrated remarkable capabilities in predicting complex patterns and relationships within diverse datasets [93,94]. Their strength lies in their ability to capture non-linear interactions among various process parameters—a challenge that conventional methods often struggle to address. By leveraging extensive datasets and sophisticated algorithms, machine learning models can unveil subtle correlations that might otherwise remain hidden. This not only enhances prediction accuracy, but also fosters a comprehensive understanding of the underlying factors shaping surface roughness. Moreover, the adaptability of machine learning techniques contributes to their predictive prowess [95]. Unlike rigid regression models, machine learning algorithms can continuously learn from new data, integrating fresh insights to refine their predictions. This adaptive nature proves especially advantageous in the realm of additive manufacturing, where process conditions and material properties can vary substantially. Consequently, machine learning empowers the development of predictive models that evolve alongside the manufacturing process, ensuring consistently precise predictions of surface roughness.

Deep learning, a subset of machine learning, adds a layer of sophistication to predictive modeling [96,97]. Neural networks within deep learning architectures automatically extract hierarchical features from raw data, enabling them to capture intricate patterns with remarkable precision. This capability proves invaluable when dealing with the complex structures and intricate relationships intrinsic to additive manufacturing processes. Deep learning models excel in analyzing intricate spatial and temporal interactions, offering insights into how diverse process parameters impact surface roughness across various production stages. The depth and complexity of deep learning models enable them to unearth nuanced relationships that may evade traditional methodologies. The hierarchical representation of features empowers deep learning models to untangle the underlying mechanisms driving surface roughness, contributing to a deeper comprehension of the manufacturing process itself. Consequently, deep learning not only delivers enhanced predictive capabilities, but also advances fundamental knowledge within additive manufacturing holds tremendous potential to reshape quality assurance and process optimization.

Recently researchers gave great attention to AI for surface roughness prediction. Thus, this paper presents machine learning models or techniques used to predict the surface roughness of 3D printed parts. The remainder of the paper is organized as follows: Section 2 provides

a method used to collect and select published data for the review. Section 3 introduces an overview of AI-based surface roughness predictions in AM. Section 4 presents a machine learning algorithm used for surface roughness predictions of AM parts. Section 5 presents a discussion on the trends, challenges, and future direction of applying AI algorithms for the surface roughness prediction of AM components. Section 6 provides conclusions and future work.

2. Methodology

To conduct this review, a rigorous and comprehensive methodology was employed. This section outlines the key steps undertaken to ensure a systematic and transparent approach to the review process. The systematic review followed the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guidelines [98] to ensure transparent and comprehensive reporting of the review process. The PRISMA flowchart was used to illustrate the study selection process and the number of studies included and excluded at each stage.

2.1. Search Strategy

The search strategy was developed to identify relevant studies on the application of artificial intelligence for surface roughness prediction of additively manufactured components. The following databases were searched: PubMed, IEEE Xplore, ScienceDirect, MDPI, and Google Scholar. The search terms used included variations and combinations of the following keywords: "artificial intelligence", "machine learning", "deep learning", "surface roughness", "prediction", "additive manufacturing", and "3D printing". To consider all published papers related to the surface roughness prediction of additively manufactured components, the database search strategy focused on articles, proceedings, and books published in the journals listed above for the last 10 years.

2.2. Inclusion and Exclusion Criteria

The inclusion criteria used to select studies were defined as follows:

- Published materials in peer-reviewed journals or conference proceedings.
- Studies targeting the application of artificial intelligence techniques for surface roughness prediction in additively manufactured components.
- Studies that provided detailed methodology and results related to surface roughness
 prediction using artificial intelligence.
- Studies published in the English language.

The exclusion criteria for selecting studies were defined as follows:

- Studies that did not specifically address surface roughness prediction in additively manufactured components.
- Studies that focused solely on conventional manufacturing processes.
- Studies that were not available in full-text or were duplicates.
- Studies that were published in another language than English.

2.3. Inclusion and Exclusion Criteria

Two phases were used in the selection process: title/abstract screening and full-text assessment. The titles and abstracts of the identified studies are based on the inclusion and exclusion criteria. Any mismatch was resolved through discussion and consensus. In the full-text assessment phase, the reviewers thoroughly examined the selected articles to find out their eligibility for inclusion in the review.

2.4. Data Extraction and Analysis

Data extraction was done by the reviewer using a predefined data extraction form. The following information was extracted from each selected study:

Objective of the study.

- Methodology employed (artificial intelligence techniques, algorithms, datasets used).
- Key findings and conclusions related to surface roughness prediction.
- Limitations and future directions identified by the authors.
- The extracted data were then synthesized and analyzed to identify common themes, trends, and patterns across the reviewed studies.

3. AI-Based Surface Roughness Prediction Overview

In this section, an overview of artificial intelligence and its application in predicting surface roughness is presented.

3.1. Artificial Intelligence Overview

Artificial intelligence is defined as a subset of IT applications that can capture information from its environment, comprehend, learn, interpret, and derive actions based on their implemented objectives [99]. In its strictest definition, AI stands for the imitation by computers of the intelligence inherent in humans [100]. AI can be classified into three categories based on its capabilities: General AI, Narrow AI, and Super AI. General AI represents a level of intelligence that can undertake any intellectual task with efficiency comparable to that of a human. This form of AI possesses a broader range of capabilities to adapt to various tasks [101]. Narrow AI, on the other hand, refers to AI systems that are designed to perform specific tasks with intelligence. It is the most common and currently available form of AI in the field. Super AI, also known as Artificial General Intelligence (AGI), signifies a level of intelligence in AI systems that surpasses human intelligence. It encompasses the ability to perform tasks more proficiently than humans, along with cognitive properties. Super AI is considered the outcome or extension of general AI [102].

Machine Learning, a prominent subset of AI, falls under the narrow AI category. It is characterized by its task-specific nature and focuses on a limited range of functionalities. Machine Learning algorithms are designed to learn from input data and enhance their performance in specific domains such as image recognition or natural language processing [103]. Machine Learning (ML) is a subfield of AI that emphasizes the development of algorithms and models enabling machines to learn from data and make predictions and decisions without explicit programming. In other words, ML involves training algorithms on extensive datasets to identify patterns and relationships, which are then utilized for making predictions or decisions [104]. ML's primary objective is to analyze and learn from given datasets to perform tasks. It is classified into three categories: (1) supervised, (2) unsupervised, and (3) reinforcement learning. Figure 1 provides an overview of common ML approaches. In supervised learning, algorithms learn from labeled training data to predict outcomes, while unsupervised learning discovers relationships among features of interest using unlabeled data. Reinforcement learning enables the model to interact with the environment, learn, and take optimal actions that yield the highest rewards. ML finds application in regression, classification, and other tasks involving high-dimensional data, where training is based on learning from previous computations, and datasets can take various forms including audio signals [105], text [106], or images [107].

Machine learning plays a significant role in additive manufacturing, where it finds numerous applications. The integration of AI in additive manufacturing has the potential to revolutionize the field [108]. In a review by Meng et al. [109], various applications of machine learning in additive manufacturing were explored, such as process parameter optimization, part property prediction, geometric deviation closed-loop control, defect detection, quality assessment, etc. The review identified several primary areas of application machine learning for additive manufacturing (as illustrated in Figure 1).



Figure 1. Taxonomy of machine learning applications in the AM domain. (Adapted from [104] an open-access article distributed under the terms of the Creative Commons CC-BY license).

3.2. Surface Roughness Prediction

3.2.1. Definition of Surface Roughness and Its Measurements Techniques

Additively manufactured surfaces, as depicted in Figure 2, consist of profile, form, waviness, and roughness components, each with distinct origins and effects on product appearance and functionality. Waviness exposes machine vibrations, form arises from manufacturing system limitations, profile relates to layer-by-layer manufacturing, and roughness stems from printing and material removal errors. Waviness, akin to signal noise, emerges due to motion system planarity and deformations caused by weight or residual stress [110]. Process-specific factors, such as defects, thermal distortion, adhesion issues, support structure inadequacies, and post-processing deformation, also contribute to waviness [111,112].



Figure 2. General spatial frequency components of additively manufactured surfaces [112] (an openaccess article, © Int. J. Adv. Manuf. Technol., Springer) distributed under the terms of the Creative Commons CC-BY license).

Surface roughness, a crucial texture element, evaluates manufactured item quality by assessing topographical feature distribution. Diverse metrics are employed across industries, addressing uncertainty in 3D-printed product surface quality through multiple metrics for efficiency [113]. For instance, for the area surface roughness evaluation, the average area roughness (S_a) and area root mean squared height (S_q) are insensitive to measurement parameters [114]. Area height distribution skewness (S_{sk}) effectively characterizes surfaces in SLM parts. Surface roughness impacts microstructures with peaks and valleys of varying heights, which is crucial as components miniaturize [115]. As such, research focuses on defining and controlling surface roughness during manufacturing.

Surface topography evaluations based on 3D-scanned image data of sample surfaces are categorized as linear measurements denoted as "R" or aerial surface measurements denoted as "S", as per ISO 21920-2 [116] and ISO 25178-2:2021 [117] definitions. The roughness of a surface is often characterized by the area ratio, representing the real surface area relative to the ideal area of a smooth surface. Common height-based metrics, derived from Equations (1)–(10) in Table 3, describe surface roughness using linear profiles. Alternatively, area roughness parameters detailed in Table 4 capture roughness variations across surfaces.

Table 3. Surface roughness	metrics based on	linear measurement	[112,116]
----------------------------	------------------	--------------------	-----------

Parameter	Description	Equation	
R_a (roughness average)	The arithmetic average of the absolute values of the roughness profile *	$R_a = \frac{1}{l_e} \int_0^{l_e} z(x)dx $	Equation (1)
R_q	Root Mean Squared of measured microscopic peaks and valleys	$R_q = \sqrt{\frac{1}{l_e} \int_0^{l_e} z^2(x) dx }$	Equation (2)
R_t (total height of profile)	The vertical distance between the maximum profile peak height and the maximum profile valley depth along the evaluation length	Rt = max(z(x)) - min(z(x))	Equation (3)
R _{sk} (skewness)	Positive skewness indicates that the surface is made up of peaks and asperities, whereas negative Rsk refers to dominant valleys on the surface	$R_{sk} = rac{1}{R_q^3} rac{1}{l_e} \int_0^{l_e} z^3(x) dx$	Equation (4)
R_{ku} (kurtosis)	A measure of the sharpness of profile peaks	$R_{ku} = \frac{1}{R_q^4} \frac{1}{l_e} \int_0^{l_e} z^4(x) dx$	Equation (5)
RzDIN	The average distance of peaks to valleys (German Standard)	$R_{zJIS} = \frac{1}{s} \sum_{I=1}^{s} R_{ti}$	Equation (6)
R _z JIS	The average distance of peaks to valleys (Japanese Standard)	$R_{zJIS} = \frac{1}{5} \sum_{I=1}^{5} R_{ti}$	Equation (7)
η	Asperity-peak density *	$\eta = rac{rac{m_4}{m_2}}{6\pi\sqrt{3}}$	Equation (8)
ρ	Asperity-peak radius	$ ho=0.375\sqrt{rac{\pi}{m_4}}$	Equation (9)
σ_s	The standard deviation of asperity-peak heights *	$\sigma_{s}=\sqrt{1-rac{0.8968}{lpha}}\sqrt{m_{0}}$	Equation (10)
	* X = {x \in R , 0 \leq x \leq l _e }, m ₀ = AVG(Z ²), m ₂ = AVG($\left(\frac{d}{d}\right)$	$\left(\frac{Z}{x}\right)^2$, $m_4 = AVG(\left(\frac{d^2Z}{dx^2}\right)^2)$, $\propto =$	$=\frac{m_0m_4}{m_2^2}.$

Various methods can be employed to measure surface roughness, depending on different definitions [11,118]. In additive manufacturing research, the Ra parameter is commonly used to measure surface roughness due to its simplicity, though it lacks sensitivity to wavelength variations. As depicted in Figure 3, the shaded area is summed and divided by the length (L) to determine the surface roughness. Studies by Li et al. [119] show that the peak-to-valley distance parameter (R_z) outperforms R_q and R_a in correlating with the tactile and visual assessments of surface roughness. However, relying solely on R_z may not fully encompass the impact of appearance factors such as texture and color on human perception and surface quality. Extracting roughness profiles lacks reproducibility due to instrument dependency and operational factors, as reported in studies [21,120]. While 2D stylus measurements remain popular, non-contact 3D optical profilometry and X-ray CT scanning (ISO 25178-2:2021 [117]) are gaining traction for providing comprehensive information without surface damage.

Parameter	Description	Equation	
S _a	Deviations in the height of the surface points concerning the Mean Reference Plane of the measurement area (A)	$Sa = \frac{1}{A} \iint A Z(x,y) $	Equation (11)
S_z	Sum of the largest peak height value and the largest pit depth value within the defined area	Sz = max(z(x, y)) + min(z(x, y))	Equation (12)
Sq	Root mean square surface height	$S_q = \sqrt{\frac{1}{A} \iint A^{z^2}(x, y) dx dy}$	Equation (13)
S _{sk}	The skewness of the surface	$S_{sk} = \frac{1}{s_q^3} \frac{1}{A} \iint A^{z^3}(x, y) dx dy$	Equation (14)
S _{ku}	The kurtosis of the surface	$S_{ku} = \frac{1}{s_q^4} \frac{1}{A} \iint A^{z^4}(x, y) dx dy$	Equation (15)
RRP	The reduction in surface roughness	$RRP = rac{S_a^i - S_a^f}{S_a^i}$	Equation (16)

Table 4. A	Area rougl	hness	parameters	[2]	1,1	12,	12	1
					'			- 1



Figure 3. Calculation of surface roughness using surface profile (Ra) [11] (an open-access article (Sensors, MDPI) distributed under the terms of the Creative Commons CC-BY license).

3.2.2. AI-Based Prediction of Surface Roughness

Figure 4 illustrates the general approach to develop data-driven predictive models that can be used for surface roughness estimation. It illustrates the presence of input, which comprises condition monitoring data, along with predictive modeling, ultimately resulting in the output of surface roughness. These input data, either independently or in combination, are employed to train machine learning models for predicting surface roughness in additively manufactured components. It is crucial to emphasize that both the quality and quantity of the data significantly impact the accuracy and reliability of the predictions. Therefore, the careful collection and processing of data are vital to prevent any potential biases or errors.



Figure 4. Data-driven predictive modeling.

The prediction of surface roughness for additively manufactured components is a fundamental application of machine learning, where data-driven modeling is used to predict the surface roughness of additively manufactured components [122]. There are several types of data that can be utilized for this prediction. Commonly used data types include:

Process parameters: These include data related to the additive manufacturing process. In LPBF, for example, parameters such as laser power, scan speed, layer thickness, and hatch spacing [123] have been used for predicting surface roughness. In FDM, parameters such as the feed rate to flow rate, layer thickness, and extruder temperature have been utilized [17]. These parameters can directly impact the surface roughness of the final product.

Material properties: These include data related to the material used in the additive manufacturing process, such as powder size, particle shape, and chemical composition, especially in methods such as L-PBF for metal AM. Material properties can also have a significant impact on surface roughness [124].

Geometrical features: These include data related to the geometry of the final product, such as the angle and direction of the surface, the size and shape of the features, and the number of layers. Geometrical features can influence the surface roughness by affecting the thermal and mechanical properties of the material. For example, in FDM, building orientation has been employed to predict surface quality [125]. In the case of EBM, on the other hand, printed components, the sloping angle and surface orientation have been used to predict surface roughness [126].

Environmental conditions: These include data related to the ambient conditions during the additive manufacturing process, such as temperature, humidity, and pressure. These conditions can affect the material properties and, in turn, the surface roughness of the final product [72].

Surface texture data: These types of data involve measuring the surface roughness of an additively manufactured component using specialized instruments such as profilometers or surface roughness testers. The surface texture data can be used as a training dataset for machine learning models to predict surface roughness based on process parameters, material properties, geometrical features, and environmental conditions [34,127].

Vibration data: Vibration data play a crucial role in predicting surface roughness for additively manufactured components. By capturing and analyzing the vibrational characteristics during the additive manufacturing process, valuable insights can be gained regarding the quality and surface roughness of the manufactured components. Vibration data can help identify any anomalies, such as excessive vibrations or oscillations that may affect the surface roughness. In the FDM-based 3D printing process, for instance, vibration data were integrated to monitor the vibration of the extruder and the table [67]. Vibration data can also be integrated into the machine learning models to enhance the accuracy of surface roughness predictions and further optimize the additive manufacturing process.

4. AI-Based Surface Roughness Prediction Methods for Additively Manufactured Parts

In this section, we explore the literature of traditional and deep learning and machine learning algorithms. These incredible algorithms have been used to uncover the secrets of predicting surface roughness in components made through additive manufacturing.

4.1. Traditional Machine Learning Approaches

It should be emphasized that conventional machine learning techniques offer a strong basis for surface roughness prediction. However, they might face challenges in capturing intricate non-linear connections or managing data with high dimensions. The application of deep learning techniques has gained prominence in recent years due to their ability to handle these challenges more effectively. The common traditional machine learning algorithms that can be applied for the surface roughness prediction of additively manufactured components are Linear Regression, Support Vector Machine, Random Forests, Gradient Boosting Algorithms, and Artificial Neural Networks.

4.1.1. Support Vector Machines

A surface roughness prediction in the context of additive manufacturing can be facilitated through various supervised learning algorithms. One such algorithm is SVM, which aims to find an optimal hyperplane for data classification. SVM is widely used due to its ability to handle both linear and nonlinear data, as well as its high accuracy [128].

Singh et al. [129] applied SVM techniques to model the Wire Electrical Discharge Machining (WEDM) process of AA6063 for armor applications. They considered four input variables: pulse-on-time (Pon), pulse-off-time (Poff), servo-voltage (V S), and peak-current (IP), with surface roughness as the response parameter. By employing a 3k full factorial design for the experimental runs, the developed model demonstrated its predictive capability and suitability for smart manufacturing. The surfaces of the machined components were further evaluated using SEM analysis.

Joshi et al. [130] explored the applications of supervised learning algorithms, specifically Support Vector Machines and Random Forests, for the surface roughness prediction of FDM-printed parts. Support Vector Machines provide high accuracy in data classification and were used to determine whether the final parts met the desired surface roughness specifications. Random Forests, an ensemble of decision trees, were employed for classification and regression tasks. The study reviewed the implementation of these algorithms and analyzed their applications in additive manufacturing.

Mishra et al. [131] focused on predicting the surface roughness of FDM-printed polylactic acid (PLA) specimens using various supervised machine learning regression-based algorithms. The study utilized explainable AI techniques to enhance the interpretability of the machine learning models. Algorithms such as Support Vector Regression, Random Forest, XGBoost, AdaBoost, CatBoost, the Decision Tree, the Extra Tree Regressor, the Explainable Boosting Model (EBM), and the Gradient Boosting Regressor were employed. The experimental results showed that the XGBoost algorithm achieved the highest coefficient of the determination value (0.9634), indicating its superior accuracy in predicting surface roughness compared to other algorithms. The study also provided a comparative analysis of algorithm performance and insights derived from explainable AI techniques.

4.1.2. Random Forests

Random forests are a type of ensemble model that combines multiple decision trees. Each tree is trained on a different subset of data and features, and the final prediction is obtained through a voting or averaging process. The strength of random forests lies in their ability to handle complex relationships and interactions among input parameters, making them well-suited for surface roughness prediction [132].

Li et al. [17] utilized a random forest (RF) algorithm as part of an ensemble learningbased approach for surface roughness prediction in Fused Filament Fabrication (FFF) processes. The authors incorporated multiple sensors to gather real-time condition monitoring data and extracted a set of features from the raw sensor-based signals in both the time and frequency domains. To enhance computational efficiency and mitigate overfitting, the researchers employed RF to select a subset of 40 features based on their importance. The ensemble learning algorithm in this study combined six different machine learning algorithms, including RF, AdaBoost, CART, SVR, RR, and the RVFL network. The experimental results demonstrated the predictive models' capability to accurately predict the surface roughness of 3D-printed specimens. The ensemble model outperformed the individual base learners, as evidenced by the lower Root Mean Square Error (RMSE) and Relative Error (RE). The authors concluded that this ensemble learning-based approach, incorporating RF as a feature selection method, shows promise for predicting the surface roughness of additively manufactured components in other processes such as selective laser sintering and electron beam melting. In their article, Wu et al. [72] addressed the need for real-time process monitoring and predictive modeling in additive manufacturing, with a specific focus on surface roughness prediction in Fused Deposition Modeling (FDM). While existing research has predominantly emphasized physics-based approaches, this study presents a novel data-driven approach using random forests for surface roughness prediction. To achieve this, a real-time monitoring system is developed, integrating multiple sensors to monitor the health condition of a 3D printer and the FDM processes. The RF algorithm is employed to construct a predictive model based on the collected sensor data. The experimental results demonstrate the high accuracy of the predictive model in estimating the surface roughness of printed parts.

Wu et al. [122] utilized random forests to train a model for predicting the surface roughness of parts produced by the FDM process. The study employed a feature-level fusion process to extract features from multiple sensor data, including the temperature of the table and extruder, vibrations of the table and extruder, etc., and combined them into a single feature vector. This vector served as the input to the model which used regression trees to predict the surface roughness value. The study also generated models with individual sensor inputs; however, sensor fusion proved to be a more accurate method, achieving the lowest cross-validation error of 5.91%.

Machine learning algorithms, including random forests, support vector regression, ridge regression (RR), and a least absolute shrinkage and selection operator (LASSO), were employed by Wu et al. [72] to accurately predict surface roughness in manufacturing parts.

4.1.3. K-Nearest Neighbors

K-nearest neighbors (KNN) is a non-parametric algorithm that utilizes proximity to the nearest neighbors to classify or predict data points. In the context of surface roughness prediction, KNN can estimate roughness by identifying the nearest neighbors with similar input parameter values [133].

Kumar and Jain [34] focused on employing the KNN machine learning algorithm for surface roughness prediction. The authors generated training data for the KNN algorithm by depositing multi-layer single-track depositions, resulting in wall-like structures, using Stellite-6 as the additive manufacturing material in both powder and wire forms. Their findings revealed that surface roughness increases with a higher power supply to the microplasma and AM material feed rate, while it decreases with an increase in the traverse speed of the deposition head for both powder and wire forms of the AM material. Additionally, the surface roughness of the walls produced with the powder form of the AM material (ranging from 118 to 149 μ m) was found to be smaller than that obtained with the wire form (ranging from 195 to 227 μ m). The prediction error of surface roughness using the KNN algorithm ranged from -6.2% to 2.8% for the powder form and -5.8% to 2.3% for the wire form of the AM material. These results demonstrate the capability of the KNN algorithm in accurately predicting surface roughness in the μ -PTAMAM process. Furthermore, the authors highlighted that increasing the number of training datasets can further reduce the prediction error of the KNN algorithm.

In the paper [134], the focus is on accurately predicting the surface finish of fused deposition modeling parts using prediction models. Previous models have successfully established a mapping relationship between printing parameters and surface roughness, though they are limited in their ability to handle multi-factor and multi-category predictions and imbalanced data. To address these challenges, a new prediction method called APSO-KNN (adaptive particle swarm optimization and K-nearest neighbor) is proposed. The method considers seven input variables: nozzle diameter, layer thickness, number of perimeters, flow rate, print speed, nozzle temperature, and build orientation. An L27 Taguchi experimental design is utilized to determine the printing values for each specimen. The model is trained and validated using experimental data from 27 printed specimens. The results demonstrate that the proposed method achieves a minimum classification error of 0.01 after two iterations, with a maximum accuracy of 99.0% and high

training efficiency. This method successfully meets the requirements for predicting the surface finish in FDM parts with multiple factors and categories, while also addressing imbalanced data. The high accuracy attained by the model indicates its potential for predicting the surface finish and its applicability in actual industrial manufacturing processes.

4.1.4. Artificial Neural Networks (ANN)

ANNs are fascinating machine learning algorithms inspired by the intricate structure and functionality of the human brain. Composed of interconnected nodes called "neurons", ANNs possess the remarkable ability to process and transmit information. These networks find applications in a wide range of tasks, including pattern recognition, image and speech recognition, prediction, and classification [135].

One captivating application of ANNs lies in the prediction of surface roughness for additively manufactured components [136]. Additive manufacturing, a process that builds 3D objects layer by layer using computer models, presents a challenge in controlling the surface roughness of printed objects—a crucial factor affecting their mechanical properties and overall performance. To address this challenge, ANNs can be trained on datasets comprising input variables and corresponding surface roughness measurements. These input variables encompass factors such as the type of printing material, layer thickness, printing speed, and printing temperature. As ANNs learn from the data, they identify patterns and make predictions based on new inputs. Once trained, ANNs can accurately predict the surface roughness of new additively manufactured components based on their input parameters. This predictive capability empowers manufacturers to optimize their printing processes and enhance the quality of their printed parts [137].

ANN have been widely utilized by researchers to predict the surface roughness of additively manufactured components. In a remarkable study conducted by Wafa and Abdulshahed [138], they employed an ANN approach to accurately predict the surface roughness of FDM-printed components. The ANN model, built with a small number of neurons in the MATLAB environment, exhibited exceptional agreement with the experimental data, achieving an average error value of only 8%. Furthermore, the researchers compared their proposed ANN model to a regression-based approach and found that the ANN model outperformed the statistical method in terms of accuracy.

In another intriguing work, Lakshmi and Arumaikkannu [139] suggested a non-contact method to estimate surface roughness in SLM-customized implants using an ANN. The developed ANN was trained on scan data from a femur bone and achieved an impressive prediction accuracy of 97.2%.

Vahabli and Rahmait [140] made significant advancements by developing a robust model for estimating the surface roughness distribution in FDM parts. They used empirical data and optimized ANN, incorporating particle swarm optimization (PSO) and imperialist competitive algorithm (ICA) techniques. The comprehensive validation of their proposed methodology showcased a superior performance across various conditions and complex shapes, effectively filling a crucial gap in the literature.

The work of Mahapatra and Sood [141] focused on the relationship between input factors and surface roughness using ANN. They applied the Levenberg–Marquardt algorithm to train the neural network, revealing valuable insights regarding the impact of factors of FDM such as raster angle, raster width, layer thickness, orientation, and air gap on surface roughness. In another investigation, Vahabli and Rahmati [142] utilized the radial basis function neural network (RBFNN) to estimate surface roughness. By optimizing the effective parameters using the imperialist competitive algorithm, they achieved improved prediction accuracy compared to using RBFNN alone for FDM parts.

Jiang et al. [143] investigated the response of the FDM process and printable bridge length (PBL) in test specimens. Their study utilized the BP neural network with Bayesian Regularization as the training algorithm, leading to accurate predictions of a test specimen's PBL. Barrios and Romero [144] deployed decision-tree methods, such as J48, random forest, and random tree, to predict the surface roughness of FDM parts. Through a welldesigned dataset and utilizing the Taguchi design and response surface methodology, they successfully screened and optimized surface roughness.

Plaza et al. [145] conducted a study on the effect of build orientation, feed rate, and layer thickness on the surface roughness of workpieces fabricated from PLA filament. Their factorial design method revealed that the layer thickness significantly impacted surface roughness, while the feed rate had no significant effect. They also employed an ANN, utilizing feedforward BP with the Levenberg–Marquardt training algorithm, to predict the surface roughness of FDM printed accurately.

Pfleging [146] aimed to predict and optimize the surface roughness and heat-affected zone depths. The data obtained were utilized to create linear regression and artificial neural network models for each variable. Subsequently, the models with the best performance were employed in a multiobjective genetic algorithm optimization to determine optimal parameter combinations. Through this approach, the research successfully identified an acceptable range of values for the specified input parameters of LPBF (laser power, focal offset, axial feed rate, number of repetitions, and scanning speed) that produced satisfactory values of Ra and HAZ simultaneously. In the pursuit of optimizing the tensile strength, surface roughness, and build time of fabricated work pieces, Yang et al. [48] employed the central composite design (CCD) method. Their study emphasized the significant impact of the nozzle diameter, filling velocity, and layer thickness on surface roughness.

Kandananond [147] utilized the Box–Behnken design to optimize the surface roughness of workpieces fabricated from ABS filament. The study successfully recommended optimal settings for input factors such as nozzle temperature, bed temperature, and printing speed, effectively minimizing surface roughness. The comparison study between the response surface method (RSM) and machine learning methods, including ANN and fuzzy inference system (FIS), provided valuable insights into the relationships between the inputs and outputs of the FDM system.

Lastly, Ulkir et al. [148] conducted a study to optimize the prediction model for minimizing surface roughness in FDM samples. They used a combination of a Cascade-Forward Neural Network (CFNN) and a genetic algorithm to determine the optimal combination of input parameters, yielding impressive results.

4.1.5. Others Machine Learnings

There are other machine learning algorithms used for surface roughness prediction. For instance, a machine learning method based on Gaussian Process Regression was proposed to establish a model relating the WAAM process parameters to the top surface roughness. To measure the top surface roughness of a manufactured part, a 3D laser measurement system was developed. The experimental datasets were collected and subsequently divided into training and testing datasets. Using the training datasets, a top surface roughness model was constructed and then verified using the testing datasets. The experimental results demonstrate that the proposed method achieves a surface roughness prediction accuracy of less than 50 μ m [149].

Li et al. [17] introduced the ensemble learning algorithm to determine predictive models for surface roughness. Their approach involved training the model using different learning algorithms, including random forests, AdaBoost, classification and regression trees (CART), SVR, RR, and random vector functional link (RVFL) networks. The experimental results demonstrated the effectiveness of this ensemble learning-based approach, particularly in accurately predicting the surface roughness of fused filament fabrication (FFF)-manufactured components.

Mishra and Jatti [92] conducted a comprehensive comparison of three quantum algorithms—Quantum Neural Network (QNN), Quantum Forest (Q-Forest), and Variational Quantum Classifier (VQC)—adapted for regression to predict surface roughness in additively manufactured components. Through the evaluation of the Mean Squared Error (MSE), Mean Absolute Error (MAE), and Explained Variance Score (EVS), the study reveals that the Q-Forest algorithm outperforms the others, achieving an MSE of 56.905, MAE of

7.479, and an EVS of 0.2957. In contrast, the QNN algorithm exhibits a higher MSE and MAE, along with a negative EVS, suggesting its limited suitability for surface roughness prediction. The VQC adapted for regression also demonstrates an inferior performance compared to the Q-Forest algorithm.

4.2. Deep Learning Approaches

Machine learning techniques are considered a weak AI type, which means they are not entirely autonomous and require some level of guidance, such as adjusting hyperparameters. Deep learning (DL) methods were developed to push beyond the limitations of traditional machine learning, which are also aimed to emulate certain aspects of human cognition more closely. Consequently, they have proved to have a superior performance in terms of accuracy and speed compared to other machine learning algorithms, without the need for significant manual intervention from programmers. Deep learning is a specific subset of machine learning that operates by processing inputs through a biologically inspired ANN architecture. Over time, it has become evident that neural networks outperform many other algorithms in accuracy and speed due to their powerful ability to extract relevant information from vast amounts of data [150].

As can be observed from Figure 5, the main difference between deep learning and traditional machine learning lies in the architecture and representation of data. In traditional machine learning, feature engineering is a separate step where domain experts manually select and engineer relevant features from the raw data. While traditional methods may be more interpretable, they heavily rely on domain knowledge and might struggle to capture intricate relationships or patterns in the data without comprehensive feature engineering. Moreover, the feature engineering process can be time-consuming and subjective. However, in deep learning, feature extraction and prediction are performed in an end-to-end manner using neural networks. Raw data are directly fed into the network, and the model automatically learns hierarchical representations and complex features, eliminating the need for manual feature engineering. The neural network's layers capture abstract patterns and relationships within the data, enabling it to predict surface roughness directly from the raw input, making the process more efficient and accurate with a large amount of labeled data. However, the black-box nature of deep learning models may limit interpretability [151].



Figure 5. Traditional machine learning vs deep learning for surface roughness prediction.

Deep learning possesses remarkable capabilities for modeling and handling highly intricate non-linear relationships. Numerous variants of deep learning techniques are available, such as convolutional neural networks (CNNs), recurrent neural networks (RNNs), artificial neural networks (ANNs), and deep neural networks (DNNs). These neural networks consist of artificial neurons organized in multiple layers, where each layer communicates only with the immediately preceding and following layers (as shown in Figure 6). Information flows through the neural network from the input layer (which receives external data) to the output layer (which produces results), passing through several hidden layers in between. The number of hidden layers depends on the complexity of the problem being solved. In each hidden layer, neurons receive input signals from other neurons, process them by combining the input with their internal state, and produce output signals. Neurons are interconnected where the output of one serves as input to another, and each neuron may have multiple input and output connections. Specific weights are then assigned, forming the overall layer of the neural network. The learning process involves adjusting the network's parameters, specifically the weights of the connections, by minimizing observed errors in the network's output. Due to its versatile and self-adapting architecture, deep learning reduces the need for extensive feature engineering and can effectively identify and address challenges that may be challenging to detect using other techniques. However, training artificial neural networks demands a substantial amount of data for accurate learning and is computationally expensive due to the vast number (millions or more) of parameters that require optimization during the training process.



Figure 6. Schematic representation of a deep neural network, a type of artificial neural network featuring several hidden layers of neurons between the input and output layers.

Deep learning methods commonly used for surface roughness prediction of additively manufactured components include conventional neural networks, deep neural networks, generative adversarial networks, auto encoders, and deep reinforcement learning. These methods are briefly explained below.

4.2.1. Convolutional Neural Networks

Convolutional Neural Networks (CNNs) are widely used in image recognition and classification tasks, particularly in the analysis of surface topography scans and visual representations of additively manufactured components. These networks utilize convolutional layers to extract hierarchical features from input data, allowing them to capture intricate patterns and textures related to surface roughness. Researchers have recently employed CNNs for predicting the surface roughness of additively manufactured components [152].

In a recent study [153], a deep learning CNN model was utilized to predict the surface roughness of additively manufactured components. The study proposed a combined approach of CNN classification and electrical discharge-assisted post-processing to enhance the surface quality of these components. By categorizing the surface, the depth and number of polishing passes were determined. The study revealed that polishing under a low-energy regime outperformed high-energy regimes, resulting in a significant 74% improvement in surface finish. Additionally, lower-energy polishing reduced the occurrence of short-circuit discharges and elemental migration. The CNN model demonstrated 96% accuracy in predicting the surface condition through a five-fold cross-validation. Furthermore, the proposed approach substantially improved the surface finish from 97.3 to 12.62 μ m.

In another work [154], researchers demonstrated that CNNs could directly predict the energy required to deform gyroid and octet truss metamaterials using only optical images.

By exploiting the tiled nature of engineered lattices, the small dataset was augmented by subdividing the original image into smaller sub-images. During testing, the predictions from these sub-images were combined to estimate the deformation work of the entire lattice. This approach provided a rapid and cost-effective screening tool for predicting properties of 3D-printed lattices, surpassing mere inspection, and accurately estimating performance metrics, rather than just detecting defects.

In a paper [155], an automated quality grading system for the additive manufacturing process was proposed, employing a deep CNN model. The CNN model was trained offline using images of internal and surface defects in the layer-by-layer deposition of materials and tested online to detect and classify failures in the AM process at different extruder speeds and temperatures. The model exhibited an accuracy of 94% and specificity of 96%, with an F-score, sensitivity, and precision measures above 75% for classifying the quality of the printing process into five grades in real-time. The proposed online model introduced automated, consistent, and non-contact quality control, eliminating the need for the manual inspection of completed parts. The quality monitoring signal could also be utilized by the machine to suggest real-time adjustments in parameters and remedial actions. This predictive model serves as a proof-of-concept for any AM machine, facilitating the production of reliable parts with fewer quality issues and minimizing wastage of time and materials.

Machine learning can be leveraged to automate common or time-consuming engineering tasks using existing data. For example, design repositories can be employed to train deep learning algorithms to assess component manufacturability. However, methods for determining the suitability of design repositories for machine learning are lacking [156]. A study investigated this matter by employing "artificial" design repositories to assess the impact of altering dataset properties on the precision and generalizability of neural networks trained on the data. A 3D convolutional neural network was employed to estimate manufacturing metrics directly from voxel-based component geometries, focusing on additive manufacturing as a case study. The study examined three build metrics: part mass, support material mass, and build time. The results indicated that training on design repositories with a less standardized orientation and position resulted in more accurate neural networks. Furthermore, orientation-dependent metrics were more challenging to estimate than orientation-independent metrics. The convolutional neural network outperformed the baseline linear regression model for all build metrics.

Furthermore, a machine learning integrated design for the AM framework was proposed, utilizing ML's ability to learn complex relationships between design and performance spaces. A case study demonstrated ML's effectiveness in designing a customized ankle brace with a tunable mechanical performance and tailored stiffness [157].

4.2.2. Deep Neural Networks

A deep neural network (DNN) is an ANN having multiple layers between the input and output layers. A DNN consists of neurons, synapses, biases, weights, and functions. Deep neural networks have demonstrated discriminative and representative learning capabilities across various applications in recent years [158,159].

So et al. [11] developed a method to improve the quality of additively manufactured products by predicting surface roughness using data analysis techniques, including data pre-processing and DNNs combined with sensor data. The study focused on enhancing the surface roughness of the stacked wall, which is a crucial quality indicator affecting product life and structural performance. By applying data pre-processing and DNNs with sensor data, the study proposed a methodology to predict surface roughness based on process parameters. The effectiveness of the proposed methodology was validated using field data from wire + arc additive manufacturing, resulting in a mean absolute percentage error (MAPE) of 1.93%.

In the prediction of surface roughness for an AlSi10Mg aluminum alloy fabricated through laser powder bed fusion (LPBF), a deep learning framework was developed [160].

The framework involved fabricating specimens, measuring surface topography, extracting and streamlining roughness and LPBF processing data, and developing a deep neural network model. The inputs to the model were the AM process parameters, and the output was the surface profile height measurements. The deep learning framework successfully predicted the surface topography and roughness parameters for all printed specimens, with measurements well within 5% of experimental error.

In the study on selective laser melting, a deep neural network was applied to classify melt-pool images based on laser power labels [161]. The neural network demonstrated effective classification, even with melt-pool images featuring blurred edges. The classification model showed a failure rate of under 1.1% for 13,200 test images, proving its effectiveness in monitoring melt-pool images and identifying potential defects.

Post-processing methods, such as shot peening (SP), severe vibratory peening (SVP), and laser shock peening (LSP), are commonly employed to address surface imperfections and bulk defects in additive manufactured materials. In a previous study, fracture surfaces of failed samples were analyzed, and experimental data were elaborated using a machine learning-based approach [162]. The study developed a six-layer deep neural network and utilized a stacked auto-encoder (SAE) for pre-training the dataset. The ML-based model achieved prediction accuracies of over 0.96 and revealed correlations between the residual stress, hardness, surface roughness, crack initiation site depth, and fatigue life of the post-treated samples.

4.2.3. Generative Adversarial Networks

Generative Adversarial Networks (GANs) have shown promise in predicting the surface roughness of additively manufactured components by generating synthetic profiles that closely resemble real-world data. GANs consist of a generator network and a discriminator network that compete against each other, enabling them to learn the underlying distribution of surface roughness and aid in prediction tasks [163].

In a recent paper [164], a novel image processing method was proposed to enhance the quality of thermal images for feature extraction in the Directed Energy Deposition (DED)-based additive manufacturing process. The method utilized an Improved Enhanced Generative Adversarial Network (IEGAN) with a modified objective function. A penalty term was introduced to enhance the contrast ratio of the reconstructed thermal images. The effectiveness of the proposed IEGAN was demonstrated by comparing the contrast ratio with that of the original GAN. The IEGAN successfully extracted the shape of the melt pool, contributing to process monitoring in additive manufacturing.

To address the challenge of controlling porosity in metal additive manufacturing, a framework combining Generative Adversarial Networks and Mallat Scattering Transformbased autocorrelation methods was introduced [165]. This framework deconstructed the generation problem and generated individual pore geometries and surface roughness. These generated components were then stochastically reconstructed to form realizations of a porous printed part. Statistical and dimensional metrics were used to compare the generated parts with existing experimental porosity distributions.

In laser powder bed fusion (L-PBF), predicting the morphology of the melt pool is crucial for controlling the final build quality. Machine learning techniques were leveraged to predict both the quantitative attributes (size) and qualitative attributes (shape) of the melt pool [166]. An LSTM network was used to predict the area of the melt pool with high accuracy, while a Melt Pool Generative Adversarial Network (MP-GAN) synthesized images of the melt pool with high structural similarity scores. These predictions enabled the real-time monitoring and control of the melt pool for consistently better quality in L-PBF.

A novel method utilizing artificial intelligence was proposed to generate the virtual surface morphology of Ti-6Al-4V parts based on given process parameters [167]. Conditional generative adversarial networks were optimized to develop a high-resolution surface morphology image generation system. The virtual surface closely matched experimental

cases, exhibiting reduced textured microstructural behavior on the surface and decreased anisotropy in the columnar structure. This AI-guided virtual surface morphology approach can help obtain high-quality parts more cost-effectively.

In the context of quality assurance in advanced manufacturing processes such as additive manufacturing, imbalanced training data can hinder the detection of abnormal states. To address this issue, a data augmentation method based on a Generative Adversarial Network was proposed [168]. By jointly optimizing a standard GAN and a classifier, high-quality generated samples were used to provide a balanced training set, improving the performance of abnormal state detection in polymer and metal additive manufacturing processes.

4.2.4. Autoencoders

Autoencoders are a type of unsupervised learning model used for dimensionality reduction and feature extraction. They encode and decode input data to learn compact representations of surface roughness profiles. These representations can be used in traditional machine learning models or for visualization purposes [169].

In the context of additive manufacturing, post-processing methods are commonly employed to address surface imperfections and bulk defects. Previous studies analyzed the effects of different peening-based treatments on the fatigue performance of laser powder bed fusion AlSi1Mg samples. The fracture surfaces of failed samples were further analyzed, and machine learning-based approaches were utilized to identify correlations between the residual stress, hardness, surface roughness, depth of the crack initiation site, and fatigue life of the post-treated samples. A deep neural network (DNN) and stacked autoencoder (SAE) were used to develop a machine learning model, with the SAE providing accurate predicted results. Parametric analyses and sensitivity analyses were performed to assess the importance of each input factor. The results showed that enhancing surface hardening, inducing higher compressive residual stresses, and reducing surface roughness led to deeper crack initiation sites and improved fatigue life [162].

Additive manufacturing is increasingly used in quality-critical applications, such as aerospace and healthcare, for fabricating complex geometries with novel materials. However, the layer-wise surface quality in additive manufacturing can be relatively poor, affecting the properties and functionality of the products. Surface morphology studies have demonstrated the significant impact of AM machine parameters on the resulting surface. Analyzing and correlating morphology features with machine parameters pose challenges due to the highly nonlinear nature of surface profiles and the presence of outliers and missing regions. To address these challenges, a convolutional autoencoder-based approach is applied in this paper to extract informative features from surface profiles. The autoencoder model, with regularization, can effectively discover low-dimensional representations from high-dimensional surface profile data, even in the presence of corruptions. Supervised machine learning is then employed to quantify the correlation between surface morphology and machine parameters. A case study in laser engineered net shaping (LENS) validates the effectiveness of the proposed method, achieving a classification accuracy of 70%, outperforming benchmark methods. Thus, the developed convolutional autoencoderbased approach shows promise for extracting surface morphology features in additive manufacturing [170].

4.2.5. Deep Reinforcement Learning

Deep Reinforcement Learning (DRL) is a combination of deep learning and reinforcement learning techniques. In the context of surface roughness prediction in additive manufacturing, DRL has emerged as a promising approach for optimizing the manufacturing process to achieve desired surface roughness outcomes. By employing DRL agents, which are intelligent systems, optimal control policies can be learned through interaction with the manufacturing environment and feedback on surface roughness performance [171]. This study introduces a DRL framework for deriving a versatile control strategy aimed at minimizing the occurrence of defects. The control policy generated by the DRL framework modifies either the velocity or power of the laser during the melting process. This adjustment ensures the consistency of the melt pool and reduces overheating in the final product. To train and validate the control policy, efficient simulations of the temperature distribution within the powder bed layer are conducted under different laser trajectories [172].

5. Discussion

5.1. Analysis of AI Techniques Used to Predict Surface Roughness

The presented studies demonstrate a diverse range of AI techniques employed for surface roughness prediction in additively manufactured components. Each study offers unique insights into the efficacy of different methodologies, shedding light on the advantages and limitations of AI-driven approaches in this domain. Table 5 lists a summary of some selected works on the data used as the input, the models developed, and the results obtained from the surface roughness prediction models.

Authors	AM	Data Used as an Input	The Model Used/Developed	Remarks/Results
Mishra et al. [131]	FDM/FFF	The input parameters of FDM techniques used for the study were layer height, wall thickness, infill density, infill pattern, nozzle temperature, bed temperature, print speed, and fan speed.	Support Vector Regression, Random Forest, XGBoost, AdaBoost, CatBoost, Decision Tree, the Extra Tree Regressor, the Explainable Boosting Model (EBM), and the Gradient Boosting Regressor.	The XGBoost algorithm outperforms the other algorithms with the highest coefficient of determination value of 0.9634. This provides the most accurate predictions for surface roughness compared with other algorithms.
Xia et al. [173]	WAAM	Data from the WAAM process parameters, specifically Welding speed (mm/s), Wire feed speed (m/min), and Overlap ratio were collected using a full factorial design.	ANFIS, ELM, and SVR machine learning models were developed to predict the surface roughness.	The comparison results indicate that GA-ANFIS has superiority in predicting surface roughness. The RMSE, R ² , MAE, and MAPE for GA-ANFIS were 0.0694, 0.93516, 0.0574, and 14.15%, respectively.
Huang et al. [134]	FDM	Process parameters of nozzle diameter, layer thickness, number of perimeters, flow rate, print speed, nozzle temperature, and build orientation were used.	The model which combines the adaptive particle swarm optimization and K-nearest neighbor (APSO-KNN) algorithms was developed.	The results indicate that the proposed method can achieve a minimum classification error of 0.01 after two iterations, with a maximum accuracy of 99.0%, and high model training efficiency.
Saxena et al. [91]	FDM	Five parameters that influence layer geometries: layer height, infill density, printing speed, and nozzle temperature with a 0-degree raster angle were used.	Support vector machine, linear regression, and two ensemble learning techniques, Xtremegradient boosting and random forest regressor model, were used for prediction of surface roughness.	By applying all machine learning algorithms, random forest regression is the best model, which gives 94.85% accurate results in datasets with a minimum mean squared error of approximately 0.1255 and a maximum r2 score of approximate. 0.9685.
Kandananond [174]	FFF	Bed temperature (80, 85, and 90 °C), printing speed (40, 80, and 120 mm/s), and layer thickness (0.10, 0.25 and 0.40 mm).	Box-Behnken and ANN	The application of ANN to accurately predict surface roughness is also recommended in this study.
Singh et al. [129]	WEDM	Four input variables, namely pulse-on-time (Pon), pulse-off-time (Poff), servo-voltage (VS), and peak-current (IP) were used as data.	Support Vector Machine Technique	The correlation value of the SVM model was 0.9634. The maximum value of roughness, i.e., 6.777 μ m, was achieved for Pon = 125 μ s, Poff = 60 μ s, IP = 30 A, and VS = 40 V; while, the minimum value of roughness, i.e., 2.381 μ m, was obtained for Pon = 105 μ s, Poff = 40 μ s, IP = 150 A, and VS = 60 V. Close to 65% improvement in surface finish value was achieved owing to a suitable setting of machining parameters.
Wafa and Abdulshahed [138]	FDM	Nozzle temperature (°C), layer height (μ m), printing speed (mm/s), Nozzle diameter (mm), and Infill density (%) considered as input variables were used.	Artificial Neural Networks (ANN)	Results show that the proposed model has high accuracy in comparison to the statistical approach. Therefore, we can use the ANN model to predict the part surface roughness for 3D printing technology. The results obtained with the proposed ANN model are also superior to the regression-based model.
Wu et al. [72]	FDM	The condition monitoring data include (1) the temperature of the build plate, (2) the temperature of the extruder, (3) the vibration of the build plate, (4) the vibration of the extruder, and (5) the temperature of the deposited material.	The predictive models are trained using random forests (RFs), support vector regression (SVR), ridge regression (RR), and least absolute shrinkage and selection operator (LASSO).	The experimental results have shown that the predictive models trained by the machine learning algorithms on the condition monitoring data predict the surface roughness of additive manufacturing parts with very high accuracy.
Li et al. [17]	FFF	Various types of sensors, such as thermocouples, infrared temperature sensors, and accelerometers, are employed to gather data on temperature and vibration.	The ensemble learning algorithm combined six different machine learning algorithms, including RF, AdaBoost, CART, SVR, RR, and RVFL network, were used.	The experimental results have shown that the predictive models are capable of predicting the surface roughness of the 3D-printed specimens. The performance of the ensemble outperforms the individual base learners based on RMSE and RE.

Table 5. The data used, the model developed, and the results of some selected surface roughness prediction models.

Table 5. Cont.

Authors	AM	Data Used as an Input	The Model Used/Developed	Remarks/Results
So et al. [11]	(GMAW)-CMT WAAM	Dynamic process parameters (travel speed, feed rate), Point clouds of bead shape.	Deep Neural Networks (DNN).	Proposed DNN model achieves 98% prediction accuracy, outperforming regression and SVR models. Uses dynamic parameters and bead shape for surface roughness prediction.
Koo et al. [175]	Powder Bed Fusion with Laser Beam (PBF-LB)	Laser power, Scanning speed, Hatching distance, Overhang angle.	Support Vector Regression (SVR), Random Forest (RF), Multilayer Perceptron (MLP).	Machine learning algorithms (SVR, RF, MLP) applied to predict downskin surface roughness in PBF-LB process. Input variables include process parameters and overhang angle. RF model demonstrated most promising results. Average deviations for SVR, RF, and MLP were 13.7%, 4.3%, and 22.5%, respectively, compared to actual measured roughness. RF showed highest prediction accuracy without using sensors.
Kumar and Jain [34]	µ-PTAMAM (Micro-Plasma Transfer Arc Metal Additive Manufacturing)	Power supply, AM material feed rate, Traverse speed.	K-nearest neighbors (KNN).	KNN predicts surface roughness in μ -PTAMAM. Training data from Stellite-6 in powder and wire forms. Powder form has lower roughness (118–149 μ m) than wire (195–227 μ m). KNN error: -6.2% to 2.8% (powder), -5.8% to 2.3% (wire).

5.1.1. Data Input Variability

The variety of input data used across these studies highlights the adaptability of AI techniques to different additive manufacturing processes. Mishra et al. [131] utilized a comprehensive set of FDM parameters, while Xia et al. [173] focused on process parameters specific to the WAAM process. The success of these studies underscores the potential for AI models to handle various input types, capturing intricate relationships between parameters and surface roughness.

5.1.2. Model Performance and Accuracy

The performance metrics across the studies, such as R2, RMSE, and MAE, reflect the success of AI models in capturing complex patterns and providing accurate surface roughness predictions. The notable performance of the XGBoost algorithm in the study by Mishra et al. [131] highlights its suitability for handling a multitude of FDM parameters, showcasing its ability to learn and model complex interactions. Similarly, the GA–ANFIS model introduced by Xia et al. [173] exhibited a superior predictive performance. The use of genetic algorithms to optimize ANFIS parameters contributed to accurate predictions, showcasing the power of hybrid models in enhancing accuracy.

5.1.3. Interpretability vs. Black Box Models

While deep learning techniques, such as ANNs and XGBoost (https://xgboost.ai/), consistently demonstrated high accuracy, they often sacrifice interpretability. In contrast, ensemble methods such as the Random Forest Regressor, as seen in Saxena et al. [91], and the GA–ANFIS model in Xia et al. [173], offer a balance between accuracy and interpretability. This trade-off is significant, particularly in industries, where understanding the underlying factors influencing predictions is crucial for process optimization and decision making.

5.1.4. Application to Diverse Manufacturing Processes

The versatility of AI models is evident in their application to various additive manufacturing processes. Studies such as Huang et al. [134] demonstrate the feasibility of AI techniques in predicting surface roughness for FDM, while Singh et al. [129] applied Support Vector Machine (SVM) techniques to the domain of WEDM. This showcases the potential for AI models to be tailored to different manufacturing processes, enabling their wide-ranging adoption.

5.1.5. Optimizing Manufacturing Processes

The studies collectively underline the potential of AI techniques in optimizing additive manufacturing processes. These models have the capability to predict surface roughness accurately, guiding engineers to adjust parameters for desired outcomes without extensive trial and error. This optimization potential can lead to enhanced manufacturing efficiency, reduced material waste, and improved product quality.

5.1.6. Weakness and Strength of each Developed Model

Each developed model possesses unique strengths and limitations when predicting the surface roughness of additively manufactured components. Considering the diversity of data types, a comparative analysis of the strengths and weaknesses was conducted for both machine learning models and deep learning models employed in surface roughness prediction. The strengths and weaknesses of the machine learning models were presented in Table 6, while those of the deep learning models were outlined in Table 7.

5.2. Challenges and Limitations

The literature on the application of AI techniques for surface roughness prediction in additive manufacturing highlights various limitations and challenges that must be addressed to advance this field.

Technique	Strengths	Weaknesses
Support Vector Machines (SVM)	 Effective in high-dimensional spaces Handles non-linearity through kernels Robust to outliers 	 Computationally intensive for large datasets Hyperparameter selection required Sensitive to noisy or overlapping data
Random Forests	 Handles non-linearity and interactions Robust to noisy data and outliers Feature importance 	 Prone to overfitting with complex models Limited performance on sparse data Harder to interpret with deep forests
K-Nearest Neighbors (KNN)	Simple and intuitiveCaptures complex relationshipsRobust to noise and outliers	 Sensitive to choice of neighbors (K) Computationally expensive prediction Poor performance with high dimensions
Artificial Neural Networks (ANN)	 Captures complex non-linear relationships Handles various input types Suitable for large data 	 Requires careful hyperparameter tuning Prone to overfitting with limited data Lack of transparency
Ridge Regression	 Handles multicollinearity through regularization Stable and less prone to overfitting 	 Requires tuning of regularization parameter May not perform well with non-linear relationships No feature selection

Table 6. The strengths and weaknesses of each machine learning algorithm used for surface roughnessprediction of additively manufactured components.

Table 7. The strengths and weaknesses of each deep learning algorithm used for surface roughness prediction of additively manufactured components.

Technique	Strengths	Weaknesses
Convolutional Neural Networks	Effective for image dataCaptures local patterns and featuresRobust to data variations	 Requires large labelled datasets Limited applicability to non-visual input data
Deep Neural Networks	 Versatile for various data types Captures complex relationships Handles structured/unstructured data 	 Prone to overfitting Requires extensive hyperparameter tuning
Generative Adversarial Networks	 Generates synthetic data Data augmentation potential Captures intricate patterns 	 Challenging and unstable training Mode collapse potential
Autoencoders	 Captures latent representations Dimensionality reduction Data reconstruction 	 Struggles with complex non-linear relationships Interpretability challenges
Deep Reinforcement Learning	 Sequential decision making Adaptive learning Optimal strategy discovery 	 Computationally intensive training Reward function design challenges Training instability and variance

One significant limitation is the scarcity of publicly available datasets containing comprehensive surface roughness measurements. Many studies rely on proprietary or limited experimental data, making it challenging to develop and evaluate AI models. The lack of standardized datasets hampers the ability to compare and benchmark different approaches. Additionally, the limited availability of diverse data restricts the training of AI models in a wide range of additive manufacturing scenarios, limiting their ability to

generalize. This scarcity of high-quality datasets poses a challenge in achieving accurate predictions of surface roughness using AI models. Indeed, the scientific community is making continuous efforts to tackle the surface roughness issue in additive manufacturing. For instance, researchers are actively exploring methods such as supervised machine learning regression-based algorithms to predict the surface roughness of additive-manufactured polylactic acid specimens [131]. Additionally, data-driven predictive modeling approaches are being developed to enhance surface roughness prediction in additive manufacturing processes. These research endeavors aim to improve the understanding and control of surface roughness in the context of additive manufacturing, ultimately leading to the better quality and performance of the manufactured parts.

Another challenge lies in feature selection and extraction. While some studies have explored manual feature engineering techniques, these processes can be time-consuming and subjective. Furthermore, relevant feature selection depends heavily on the specific additive manufacturing process and material. Although automated feature learning using deep learning models has gained attention, determining the most informative features from complex datasets remains a challenge. Further research is necessary to develop robust and automated feature selection methods tailored specifically for surface roughness prediction in additive manufacturing.

The generalization of AI models across different additive manufacturing processes and materials poses another significant challenge. Each combination of process and material exhibits unique characteristics that influence surface roughness. AI models trained on specific process–material combinations may not effectively transfer to other scenarios, limiting their applicability. Developing transferable models capable of handling variations in process parameters, material properties, and surface characteristics is crucial for the broader adoption of AI techniques in predicting surface roughness. Additionally, the complexity and variability of additive manufacturing processes make it difficult to develop models that accurately predict surface roughness across different materials and printing parameters. Capturing the complex and nonlinear relationship between process parameters and the resulting surface roughness is crucial. A data-driven predictive modeling approach that utilizes multiple sensors to collect temperature and vibration data has been introduced to improve the surface integrity of additively manufactured parts.

Furthermore, the interpretability and explainability of AI models present challenges in this domain. Many AI models, especially deep learning models, are often perceived as black boxes, making it difficult to understand the reasoning behind their predictions. This lack of interpretability hampers insights into the factors contributing to surface roughness. Research efforts should focus on developing explainable AI models that shed light on the relationships between process parameters, material properties, and surface roughness. Ensuring the transparency and interpretability of these models is crucial for their successful integration into real-world manufacturing processes.

In addition to the aforementioned challenges, training challenges and computational costs also play a significant role. Training AI models for accurate surface roughness prediction demands substantial computational resources and time. The complexity of the underlying data, along with the need for large and diverse datasets, often results in prolonged training periods. Furthermore, finding an optimal balance between model complexity and generalization capacity is a challenge. Models that are overly complex might overfit to training data and perform poorly on unseen data, while overly simplified models might lack the ability to capture intricate relationships in the data.

Considering computational costs is paramount, especially in real-time applications where quick predictions are required. Complex AI models, such as deep neural networks, can demand substantial computational power for both training and inference phases. This can lead to challenges in deployment, especially in resource-constrained environments. Optimization techniques and hardware acceleration solutions are being explored to mitigate these computational challenges. Data quality and preprocessing complexity present further challenges. Acquiring precise and dependable measurements of surface roughness can be complex due to factors such as measurement equipment accuracy, noise, and variations in measurement techniques. Ensuring consistent data quality across diverse sources is essential for training reliable AI models. Additionally, the process of preprocessing raw data to eliminate outliers, address missing values, and standardize data for training can introduce its own intricacies and potential sources of error.

Moreover, integrating AI techniques into existing manufacturing processes and systems presents implementation challenges. Incorporating AI-based surface roughness prediction models into real-time control systems requires considerations such as computational efficiency, real-time data acquisition, and compatibility with existing manufacturing infrastructure. Addressing the implementation and deployment of AI models in practical manufacturing environments is vital to ensure seamless integration and practical utility.

In conclusion, while AI techniques hold promise for surface roughness prediction in additive manufacturing, several limitations and challenges must be overcome. These include the scarcity of comprehensive and standardized datasets, challenges in feature selection and extraction, limitations in model generalization, a lack of interpretability, and implementation challenges. Addressing these limitations and challenges will pave the way for more accurate, robust, and practical AI-based surface roughness predictions in additive manufacturing.

5.3. Future Direction

The existing literature on the application of AI techniques for surface roughness predictions in additive manufacturing suggests several promising future directions that can further advance this field. These directions address the limitations and challenges identified and aim to enhance the accuracy, applicability, and practicality of AI-based surface roughness predictions.

One crucial future direction is the integration of AI techniques with process optimization algorithms. By combining predictive models with closed-loop control systems, real-time adjustments can be made to process parameters during additive manufacturing. This integration enables the dynamic control of surface roughness, leading to improved surface quality control. The optimization algorithms can utilize the predicted surface roughness values to adjust process parameters, ensuring the desired surface roughness is achieved. This integration holds the potential to optimize additive manufacturing processes and improve overall manufacturing efficiency.

The utilization of generative AI models, such as generative adversarial networks (GANs), is another promising future direction. GANs can be employed to generate synthetic surface roughness patterns that resemble real-world variations. These generative models can augment limited datasets, simulate different surface roughness scenarios, and enhance the robustness of AI models. By generating synthetic data, GANs can alleviate the data scarcity issue and improve the generalization capabilities of AI models.

The development of explainable AI models is also critical for the future of surface roughness predictions in additive manufacturing. Explainability is essential for users and stakeholders to understand and trust the predictions made by AI models. Methods for explaining the decision-making process of AI models need to be explored and implemented. This can provide insights into the factors influencing surface roughness predictions and enable users to identify opportunities for process optimization and improvement. Explainable AI models can enhance transparency, trust, and acceptance of AI techniques in the additive manufacturing industry.

Currently, the majority of studies in surface roughness prediction for additively manufactured components primarily concentrate on process parameters. Only a limited number of papers explore alternative data sources such as vibration data, material properties, environmental conditions, and geometric features for predictive modeling. Future research endeavors should explore and integrate these underutilized data types to enhance the accuracy and robustness of surface roughness predictions in additive manufacturing.

Additionally, future research should focus on exploring advanced feature selection techniques specifically tailored for surface roughness prediction in additive manufacturing. The development of automated feature extraction methods that can effectively capture the underlying factors influencing surface roughness would enhance the predictive performance of AI models. Advanced feature selection techniques, such as deep feature learning or genetic algorithms, can help identify the most relevant and informative features from complex datasets.

6. Conclusions

This review paper has explored the application of AI for surface roughness predictions of additively manufactured components. The literature review has provided an overview of the recent advancements in machine learning and AI deep learning techniques employed by researchers in this field. The findings demonstrate that AI-based methods have shown promising results in predicting surface roughness, offering benefits such as cost reduction and time savings. However, certain limitations and challenges, including the availability of quality data and model interpretability, need to be addressed. Future research should focus on addressing these challenges, developing more accurate and robust models, and integrating AI methodologies into the additive manufacturing process. Overall, leveraging AI for surface roughness predictions holds great potential for improving the productivity, competitiveness, and consistent product quality of additively manufactured components. Future research should focus on addressing the challenges and refining AI-based approaches to enhance surface quality control in additive manufacturing processes.

Author Contributions: Conceptualization, T.B.; methodology, T.B.; validation, T.B., H.G.L. and H.S; resources, T.B., H.G.L. and H.S.; data curation, T.B., H.G.L. and H.S.; writing—original draft preparation, T.B.; writing—review and editing, T.B., H.G.L. and H.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Andreoni, A.; Gregory, M. Why and how does manufacturing still matter: Old rationales, new realities. *Rev. d'Economie Ind.* 2013, 144, 21–57. [CrossRef]
- Helper, S.; Krueger, T.; Wial, H. Why does manufacturing matter? Which manufacturing matters? A policy framework. SSRN 2021. [CrossRef]
- 3. Pereira, T.; Kennedy, J.V.; Potgieter, J. A comparison of traditional manufacturing vs additive manufacturing, the best method for the job. *Procedia Manuf.* **2019**, *30*, 11–18. [CrossRef]
- 4. Baumung, W. Design of an architecture of a production planning and control system (ppc) for additive manufacturing (am). *Lect. Notes Bus. Inf. Process.* **2020**, *389*, 391–402. [CrossRef]
- Framinan, J.M.; Perez-Gonzalez, P.; Fernandez-Viagas, V. An overview on the use of operations research in additive manufacturing. Ann. Oper. Res. 2023, 322, 5–40. [CrossRef]
- 6. Gardan, J. Additive manufacturing technologies: State of the art and trends. Int. J. Prod. Res. 2016, 54, 3118–3132. [CrossRef]
- 7. Gebhardt, A. Understanding Additive Manufacturing. Underst. Addit. Manuf. 2011, 2011, I–IX. [CrossRef]
- 8. Rathee, S.; Srivastava, M.; Maheshwari, S.; Kundra, T.K.; Siddiquee, A.N. *Friction Based Additive Manufacturing Technologies: Principles for Building in Solid State, Benefits, Limitations, and Applications; CRC Press: Boca Raton, FL, USA, 2018.*
- 9. ASTM52900; Additive Manufacturing—General Principles Terminology. ISO: Geneva, Switzerland; ASTM: West Conshohocken, PA, USA, 2013.
- 10. Srivastava, M.; Rathee, S.; Patel, V.; Kumar, A.; Koppad, P.G. A review of various materials for additive manufacturing: Recent trends and processing issues. *J. Mater. Res. Technol.* **2022**, *21*, 2612–2641. [CrossRef]

- 11. So, M.S.; Seo, G.J.; Kim, D.B.; Shin, J.H. Prediction of Metal Additively Manufactured Surface Roughness Using Deep Neural Network. *Sensors* **2022**, *22*, 7955. [CrossRef]
- Zhou, Y.; Chen, H.; Tang, Y.; Gopinath, S.; Xu, X.; Zhao, Y.F. Simulation and optimization framework for additive manufacturing processes. In Proceedings of the 2014 International Conference on Innovative Design and Manufacturing (ICIDM), Montréal, QC, Canada, 13–15 August 2014; Volume 17, pp. 34–40. [CrossRef]
- 13. Dinovitzer, M.; Chen, X.; Laliberte, J.; Huang, X.; Frei, H. Effect of wire and arc additive manufacturing (WAAM) process parameters on bead geometry and microstructure. *Addit. Manuf.* **2019**, *26*, 138–146. [CrossRef]
- 14. Le, V.T.; Mai, D.S.; Tran, V.C.; Doan, T.K. Additive Manufacturing of Thin-Wall Steel Parts by Gas Metal Arc Welding Robot: The Surface Roughness, Microstructures and Mechanical Properties. *Intell. Syst. Ref. Libr.* **2021**, *193*, 43–54.
- 15. Turner, B.N.; Gold, S.A. A review of melt extrusion additive manufacturing processes: II. Materials, dimensional accuracy, and surface roughness. *Rapid Prototyp. J.* 2015, 21, 250–261. [CrossRef]
- Strano, G.; Hao, L.; Everson, R.M.; Evans, K.E. Surface roughness analysis, modelling and prediction in selective laser melting. J. Mater. Process. Technol. 2013, 213, 589–597. [CrossRef]
- Li, Z.; Zhang, Z.; Shi, J.; Wu, D. Prediction of surface roughness in extrusion-based additive manufacturing with machine learning. *Robot. Comput. Integr. Manuf.* 2019, 57, 488–495. [CrossRef]
- 18. Chugunov, S.; Smirnov, A.; Kholodkova, A.; Tikhonov, A.; Dubinin, O.; Shishkovsky, I. Evaluation of stereolithography-based additive manufacturing technology for BaTiO₃ ceramics at 465 nm. *Appl. Sci.* **2022**, *12*, 412. [CrossRef]
- 19. Khodaii, J.; Rahimi, A. Improving the surface roughness in stereolithography by controlling surface angle, hatch spaces, and postcuring time. *Eng. Rep.* **2020**, *2*, 12193. [CrossRef]
- De Pasquale, G. Additive manufacturing of micro-electro-mechanical systems (MEMS). *Micromachines* 2021, 12, 1374. [CrossRef] [PubMed]
- Petzold, S.; Klett, J.; Schauer, A.; Osswald, T.A. Surface roughness of polyamide 12 parts manufactured using selective laser sintering. *Polym. Test.* 2019, *80*, 106094. [CrossRef]
- 22. Dey, A.; Yodo, N. A systematic survey of FDM process parameter optimization and their influence on part characteristics. *J. Manuf. Mater. Process.* **2019**, *3*, 64. [CrossRef]
- 23. Kelkar, A.S.; Kumbhar, N.N.; Mulay, A.V. Surface roughness measurement of parts manufactured by FDM process using light sectioning vision system. *J. Inst. Eng. India Ser.* **2018**, *C 99*, 429–433. [CrossRef]
- 24. Mekonnen, B.G.; Bright, G.; Walker, A. A study on state of the art technology of laminated object manufacturing (LOM). In *CAD/CAM, Robotics and Factories of the Future*; Mandal, D.K., Syan, C.S., Eds.; Springer: New Delhi, India, 2016. [CrossRef]
- 25. Kechagias, J. An experimental investigation of the surface roughness of parts produced by LOM process. *Rapid Prototyp. J.* 2007, 13, 17–22. [CrossRef]
- Kalita, S.J. Rapid prototyping in biomedical engineering: Structural intricacies of biological materials. In *Biointegration of Medical Implant Materials: Science and Design*; Woodhead Publishing: Sawston, UK, 2010; Volume 12, pp. 349–397.
- 27. Townsend, A.; Senin, N.; Blunt, L.; Leach, R.K.; Taylor, J.S. Surface texture metrology for metal additive manufacturing: A review. *Precis. Eng.* **2016**, *46*, 34–47. [CrossRef]
- Safdar, A.; Wei, L.Y.; Snis, A.; Lai, Z. Evaluation of microstructural development in electron beam melted Ti-6Al-4V. *Mater. Charact.* 2012, 65, 8–15. [CrossRef]
- Prototypes, A. DMLS Materials and Characteristics for Metal 3D Printing. Available online: https://www.axisproto.com/materials/dmls/ (accessed on 23 August 2023).
- 30. Europe, X. Surface Roughness In 3D Printing | Xometry Europe. Available online: https://xometry.eu/en/surface-roughness-in-3d-printing/ (accessed on 23 August 2023).
- Cavallo, C. All About Binder Jetting 3D Printing. Available online: https://www.thomasnet.com/articles/custom-manufacturingfabricating/all-about-binder-jetting-3d-printing/ (accessed on 23 August 2023).
- Huckstepp, A. (14) Surface Roughness in Metal AM | LinkedIn. Available online: https://www.linkedin.com/pulse/surfaceroughness-metal-am-alex-huckstepp/ (accessed on 23 August 2023).
- Ahn, D.G. Directed Energy Deposition (DED) Process: State of the Art. Int. J. Precis. Eng. Manuf.—Green Technol. 2021, 8, 703–742. [CrossRef]
- 34. Kumar, P.; Jain, N.K. Surface roughness prediction in micro-plasma transferred arc metal additive manufacturing process using K-nearest neighbors algorithm. *Int. J. Adv. Manuf. Technol.* **2022**, *119*, 2985–2997. [CrossRef]
- 35. de Pastre, M.A.; Quinsat, Y.; Lartigue, C. Effects of additive manufacturing processes on part defects and properties: A classification review. *Int. J. Interact. Des. Manuf.* **2022**, *16*, 1471–1496. [CrossRef]
- Li, H.; Liang, X.; Li, Y.; Lin, F. Performance of High-Layer-Thickness Ti6Al4V Fabricated by Electron Beam Powder Bed Fusion under Different Accelerating Voltage Values. *Materials* 2022, 15, 1878. [CrossRef]
- 37. Tran, H.C.; Lo, Y.L.; Yang, H.C.; Hsiao, H.C.; Cheng, F.T.; Kuo, T.H. Intelligent additive manufacturing architecture for enhancing uniformity of surface roughness and mechanical properties of laser powder bed fusion components. *IEEE Trans. Autom. Sci. Eng.* **2022**, 1–12. [CrossRef]
- Shi, X.; Ma, S.; Liu, C.; Chen, C.; Wu, Q.; Chen, X.; Lu, J. Performance of high layer thickness in selective laser melting of Ti6Al4V. Materials 2016, 9, 975. [CrossRef]

- Majeed, A.; Ahmed, A.; Salam, A.; Sheikh, M.Z. Surface quality improvement by parameters analysis, optimization and heat treatment of AlSi10Mg parts manufactured by SLM additive manufacturing. *Int. J. Light. Mater. Manuf.* 2019, 2, 288–295. [CrossRef]
- 40. Chartrain, N.A.; Williams, C.B.; Whittington, A.R. A Review on Fabricating Tissue Scaffolds using Vat Photopolymerization. *Acta Biomater.* **2018**, 74, 90–111. [CrossRef]
- 41. Nohut, S.; Schwentenwein, M. Vat Photopolymerization Additive Manufacturing of Functionally Graded Materials: A Review. J. Manuf. Mater. Process. 2022, 6, 17. [CrossRef]
- 42. Tyagi, S.; Yadav, A.; Deshmukh, S. Review on mechanical characterization of 3D printed parts created using material jetting process. *Mater. Today Proc.* 2021, 51, 1012–1016. [CrossRef]
- 43. Pugalendhi, A.; Ranganathan, R.; Ganesan, S. Impact of process parameters on mechanical behaviour in multi-material jetting. *Mater. Today Proc.* **2019**, *46*, 9139–9144. [CrossRef]
- Wang, Y.; Zhao, Y.F. Investigation of Sintering Shrinkage in Binder Jetting Additive Manufacturing Process. Procedia Manuf. 2017, 10, 779–790. [CrossRef]
- 45. Chen, H.; Zhao, Y.F. Process parameters optimization for improving surface quality and manufacturing accuracy of binder jetting additive manufacturing process. *Rapid Prototyp. J.* 2016, 22, 527–538. [CrossRef]
- 46. Mostafaei, A.; Elliott, A.M.; Barnes, J.E.; Li, F.; Tan, W.; Cramer, C.L.; Nandwana, P.; Chmielus, M. Binder jet 3D printing—Process parameters, materials, properties, modeling, and challenges. *Prog. Mater. Sci.* 2020, *119*, 100707. [CrossRef]
- Miyanaji, H. Binder jetting additive manufacturing process fundamentals and the resultant influences on part quality. J. Food Eng. 2018, 39, 31–37. [CrossRef]
- Yang, L.; Li, S.; Li, Y.; Yang, M.; Yuan, Q. Experimental Investigations for Optimizing the Extrusion Parameters on FDM PLA Printed Parts. *J. Mater. Eng. Perform.* 2019, 28, 169–182. [CrossRef]
- 49. Burke, C.; Dalal, A.; Abukhalaf, A.; Noorani, R. Effects of process parameter variation on the surface roughness of polylactic acid (PLA) materials using design of experiments (DOE). *IOP Conf. Ser. Mater. Sci. Eng.* **2020**, *897*, 012003. [CrossRef]
- Jakupi, K.; Dukovski, V.; Hodolli, G. Surface Roughness Modeling of Material Extrusion PLA Flat Surfaces. Int. J. Polym. Sci. 2023, 2023, 8844626. [CrossRef]
- 51. Jaisingh Sheoran, A.; Kumar, H. Fused Deposition modeling process parameters optimization and effect on mechanical properties and part quality: Review and reflection on present research. *Mater. Today Proc.* **2020**, *21*, 1659–1672. [CrossRef]
- 52. Solomon, I.J.; Sevvel, P.; Gunasekaran, J. A review on the various processing parameters in FDM. *Mater. Today Proc.* 2020, 37, 509–514. [CrossRef]
- 53. Kristiawan, R.B.; Imaduddin, F.; Ariawan, D.; Ubaidillah; Arifin, Z. A review on the fused deposition modeling (FDM) 3D printing: Filament processing, materials, and printing parameters. *Open Eng.* **2021**, *11*, 639–649. [CrossRef]
- Fox, J.C.; Moylan, S.P.; Lane, B.M. Effect of Process Parameters on the Surface Roughness of Overhanging Structures in Laser Powder Bed Fusion Additive Manufacturing. *Procedia CIRP* 2016, 45, 131–134. [CrossRef]
- 55. Paraschiv, A.; Matache, G.; Condruz, M.R.; Frigioescu, T.F.; Pambaguian, L. Laser Powder Bed Fusion Process Parameters' Optimization for Fabrication of Dense IN 625. *Materials* **2022**, *15*, 5777. [CrossRef]
- Svetlizky, D.; Das, M.; Zheng, B.; Vyatskikh, A.L.; Bose, S.; Bandyopadhyay, A.; Schoenung, J.M.; Lavernia, E.J.; Eliaz, N. Directed energy deposition (DED) additive manufacturing: Physical characteristics, defects, challenges and applications. *Mater. Today* 2021, 49, 271–295. [CrossRef]
- 57. Mohamed, O.A.; Masood, S.H.; Bhowmik, J.L. Optimization of fused deposition modeling process parameters: A review of current research and future prospects. *Adv. Manuf.* **2015**, *3*, 42–53. [CrossRef]
- Baturynska, I.; Semeniuta, O.; Martinsen, K. Optimization of Process Parameters for Powder Bed Fusion Additive Manufacturing by Combination of Machine Learning and Finite Element Method: A Conceptual Framework. *Procedia CIRP* 2018, 67, 227–232. [CrossRef]
- 59. Cho, K.T.; Nunez, L.; Shelton, J.; Sciammarella, F. Investigation of Effect of Processing Parameters for Direct Energy Deposition Additive Manufacturing Technologies. *J. Manuf. Mater. Process.* **2023**, *7*, 105. [CrossRef]
- 60. Valizadeh, I.; Tayyarian, T.; Weeger, O. Influence of process parameters on geometric and elasto-visco-plastic material properties in vat photopolymerization. *Addit. Manuf.* **2023**, *72*, 103641. [CrossRef]
- 61. Kumaresan, R.; Samykano, M.; Kadirgama, K.; Ramasamy, D.; Keng, N.W.; Pandey, A.K. 3D Printing Technology for Thermal Application: A Brief Review. J. Adv. Res. Fluid Mech. Therm. Sci. 2021, 83, 84–97. [CrossRef]
- 62. Dini, F.; Ghaffari, S.A.; Jafar, J.; Hamidreza, R.; Marjan, S. A review of binder jet process parameters; powder, binder, printing and sintering condition. *Met. Powder Rep.* 2020, *75*, 95–100. [CrossRef]
- 63. Patpatiya, P.; Chaudhary, K.; Shastri, A.; Sharma, S. A review on polyjet 3D printing of polymers and multi-material structures. *Proc. Inst. Mech. Eng. Part C J. Mech. Eng. Sci.* 2022, 236, 7899–7926. [CrossRef]
- 64. Gülcan, O.; Günaydın, K.; Tamer, A. The state of the art of material jetting—A critical review. *Polymers* **2021**, *13*, 2829. [CrossRef] [PubMed]
- 65. Malakizadi, A.; Mallipeddi, D.; Dadbakhsh, S.; M'Saoubi, R.; Krajnik, P. Post-processing of additively manufactured metallic alloys—A review. *Int. J. Mach. Tools Manuf.* **2022**, *179*, 103908. [CrossRef]
- 66. Tan, K.L.; Yeo, S.H. Surface finishing on IN625 additively manufactured surfaces by combined ultrasonic cavitation and abrasion. *Addit. Manuf.* **2020**, *31*, 100938. [CrossRef]

- 67. Dini, A.; Neac, A.; Portoacă, A.I.; Tănase, M.; Ilinca, C.N.; Ramadan, I.N. Additive Manufacturing Post-Processing Treatments, a Review with Emphasis on Mechanical Characteristics. *Materials* **2023**, *16*, 4610. [CrossRef]
- Syrlybayev, D.; Seisekulova, A.; Talamona, D.; Perveen, A. The Post-Processing of Additive Manufactured Polymeric and Metallic Parts. J. Manuf. Mater. Process. 2022, 6, 116. [CrossRef]
- 69. Obilanade, D.; Dordlofva, C.; Törlind, P. Surface roughness considerations in design for additive manufacturing—A literature review. *Proc. Des. Soc.* 2021, *1*, 2841–2850. [CrossRef]
- 70. Chan, K.S.; Koike, M.; Mason, R.L.; Okabe, T. Fatigue life of titanium alloys fabricated by additive layer manufacturing techniques for dental implants. *Metall. Mater. Trans. A Phys. Metall. Mater. Sci.* 2013, 44, 1010–1022. [CrossRef]
- 71. Calignano, F.; Manfredi, D.; Ambrosio, E.P.; Iuliano, L.; Fino, P. Influence of process parameters on surface roughness of aluminum parts produced by DMLS. *Int. J. Adv. Manuf. Technol.* **2013**, *67*, 2743–2751. [CrossRef]
- 72. Wu, D.; Wei, Y.; Terpenny, J. Predictive modelling of surface roughness in fused deposition modelling using data fusion. *Int. J. Prod. Res.* **2019**, *57*, 3992–4006. [CrossRef]
- Du, Y.; Mukherjee, T.; Finch, N.; De, A.; DebRoy, T. High-throughput screening of surface roughness during additive manufacturing. J. Manuf. Process. 2022, 81, 65–77. [CrossRef]
- 74. Lin, W.J.; Lo, S.H.; Young, H.T.; Hung, C.L. Evaluation of deep learning neural networks for surface roughness prediction using vibration signal analysis. *Appl. Sci.* 2019, *9*, 1462. [CrossRef]
- 75. Wang, M.; Yang, C.; Li, Z.; Zhao, S.; Zhang, Y.; Lu, X. Effects of surface roughness on the aerodynamic performance of a high subsonic compressor airfoil at low Reynolds number. *Chin. J. Aeronaut.* **2021**, *34*, 71–81. [CrossRef]
- Merryisha, S.; Rajendran, P. Experimental and cfd analysis of surface modifiers on aircraft wing: A review. CFD Lett. 2019, 11, 46–56.
- 77. Spencer, A. Optimizing Surface Texture for Combustion Engine Cylinder Liners; Luleå University of Technology: Luleå, Sweden, 2010.
- 78. Hunde, B.R.; Woldeyohannes, A.D. Future prospects of computer-aided design (CAD)—A review from the perspective of artificial intelligence (AI), extended reality, and 3D printing. *Results Eng.* **2022**, *14*, 100478. [CrossRef]
- 79. Song, S.; Terziev, M.; Tezdogan, T.; Demirel, Y.K.; De Marco Muscat-Fenech, C.; Incecik, A. Investigating roughness effects on ship resistance in shallow waters. *Ocean Eng.* 2022, 270, 113643. [CrossRef]
- Macêdo, R.; Sena, M.S.; Costa e Silva, J.; Chaves, A.; da Costa, J.A.P. The Role of Surface Roughness on the electron confinement in semiconductor quantum dots. In Proceedings of the Latin America Optics and Photonics Conference Paper LM2A.19, Sao Sebastiao, Brazil, 10–13 November 2012. [CrossRef]
- Kelly, J.; Vogel, C.; Willden, R. Impact and mitigation of blade surface roughness effects on wind turbine performance. Wind Energy 2022, 25, 660–677. [CrossRef]
- 82. Hunde, B.R.; Woldeyohannes, A.D. 3D printing and solar cell fabrication methods: A review of challenges, opportunities, and future prospects. *Results Opt.* **2022**, *11*, 2023. [CrossRef]
- Avila-Sierra, A.; Zhang, Z.J.; Fryer, P.J. Effect of surface characteristics on cleaning performance for CIP system in food processing. Energy Procedia 2019, 161, 115–122. [CrossRef]
- 84. Caroline, A.; Frabetti, C.; Oliveira, J.; Vanessa, D.M.; Lionel, J. Adhesion of food on surfaces: Theory, measurements, and main trends to reduce it prior to industrial drying. *Food Eng. Rev.* **2021**, *13*, 884–901. [CrossRef]
- 85. Bovas, B.C.; Karunamoorthy, L.; Chuan, F.B. Effect of surface roughness and process parameters on mechanical properties of fabricated medical catheters. *Mater. Res. Express* 2019, *6*, 125420. [CrossRef]
- 86. Swain, S.; Rautray, T.R. Effect of Surface Roughness on Titanium Medical Implants. In *Materials Horizons: From Nature to Nanomaterials*; Springer: Berlin/Heidelberg, Germany, 2021; pp. 55–80.
- 87. Cao, L.; Li, J.; Hu, J.; Liu, H.; Wu, Y.; Zhou, Q. Optimization of surface roughness and dimensional accuracy in LPBF additive manufacturing. *Opt. Laser Technol.* **2021**, 142, 107246. [CrossRef]
- 88. Martín Fernández, F.; Martín Sánchez, M.J. Analysis of the Effect of the Surface Inclination Angle on the Roughness of Polymeric Parts Obtained with Fused Filament Fabrication Technology. *Polymers* **2023**, *15*, 585. [CrossRef]
- Mishra, P.; Sood, S.; Bharadwaj, V.; Aggarwal, A.; Khanna, P. Parametric Modeling and Optimization of Dimensional Error and Surface Roughness of Fused Deposition Modeling Printed Polyethylene Terephthalate Glycol Parts. *Polymers* 2023, 15, 546. [CrossRef]
- 90. Kopp, M.; Uhlmann, E. Prediction of the Roughness Reduction in Centrifugal Disc Finishing of Additive Manufactured Parts Based on Discrete Element Method. *Machines* 2022, *10*, 1151. [CrossRef]
- 91. Saxena, A.; Murtaza, Q.; Kumar, P. Prediction of surface roughness in additively manufactured samples in PLA + polymer material through machine learning. *Eur. Chem. Bull.* **2023**, *12*, 2113–2124.
- 92. Mishra, A.; Jatti, V.S. Quantum Machine Learning Approach for the Prediction of Surface Roughness in Additive Manufactured Specimens. April 2023. Available online: https://arxiv.org/abs/2304.13142v1 (accessed on 10 August 2023).
- 93. Murdoch, W.J.; Singh, C.; Kumbier, K.; Abbasi-Asl, R.; Yu, B. Definitions, methods, and applications in interpretable machine learning. *Proc. Natl. Acad. Sci. USA* 2019, *116*, 22071–22080. [CrossRef]
- 94. Choudhary, K.; DeCost, B.; Chen, C.; Jain, A.; Tavazza, F.; Cohn, R.; Park, C.W.; Choudhary, A.; Agrawal, A.; Billinge, S.J.; et al. Recent advances and applications of deep learning methods in materials science. *NPJ Comput. Mater.* **2022**, *8*, 59. [CrossRef]
- 95. Ivanova, K. A Review of Deep Learning Applications in Energy—Efficient Transportation Systems. *Int. J. Intell. Autom. Comput. Res.* 2022, *5*, 15–28.

- Bhat, M.; Rabindranath, M.; Chara, B.S.; Simonetto, D.A. Artificial intelligence, machine learning, and deep learning in liver transplantation. J. Hepatol. 2023, 78, 1216–1233. [CrossRef]
- 97. Taye, M. Understanding of machine learning with deep learning: Architectures, workflow, applications and future directions. *Computers* **2023**, *12*, 91. [CrossRef]
- Page, M.J.; McKenzie, J.E.; Bossuyt, P.M.; Boutron, I.; Hoffmann, T.C.; Mulrow, C.D.; Shamseer, L.; Tetzlaff, J.M.; Akl, E.A.; Brennan, S.E.; et al. The PRISMA 2020 statement: An updated guideline for reporting systematic reviews. *BMJ* 2021, 372, n71. [CrossRef] [PubMed]
- 99. Collins, C.; Dennehy, D.; Conboy, K.; Mikalef, P. Artificial intelligence in information systems research: A systematic literature review and research agenda. *Int. J. Inf. Manag.* 2021, *60*, 102383. [CrossRef]
- 100. Haroon, S.; Corien, P.; Erik, S. Mission AI; Springer International Publishing: Cham, Switzerland, 2023. [CrossRef]
- 101. Feigenbaum, M.J.; Mermin, N.D. Artificial intelligence a modern approach. Am. J. Phys. 1988, 56, 18-21. [CrossRef]
- Zhang, L.; Pan, Y.; Wu, X.; Skibniewski, M.J. Introduction to Artificial Intelligence. In *Lecture Notes in Civil Engineering*; Springer International Publishing: Cham, Switzerland, 2021; Volume 163, pp. 1–15.
- 103. Murphy, R.R. Introduction to AI Robotics; MIT Press: Cambridge, MA, USA, 2014; pp. 1-487.
- 104. Liu, Y.; Zhao, T.; Ju, W.; Shi, S. Materials discovery and design using machine learning. J. Mater. 2017, 3, 159–177. [CrossRef]
- 105. Li, Z.; Liu, R.; Wu, D. Data-driven smart manufacturing: Tool wear monitoring with audio signals and machine learning. *J. Manuf. Process.* **2019**, *48*, 66–76. [CrossRef]
- Cheng, M.Y.; Kusoemo, D.; Gosno, R.A. Text mining-based construction site accident classification using hybrid supervised machine learning. *Autom. Constr.* 2020, 118, 103265. [CrossRef]
- Shi, B.; Bai, X.; Yao, C. An end-to-end trainable neural network for image-based sequence recognition and its application to scene text recognition. *IEEE Trans. Pattern Anal. Mach. Intell.* 2017, 39, 2298–2304. [CrossRef] [PubMed]
- Wang, C.; Tan, X.P.; Tor, S.B.; Lim, C.S. Machine learning in additive manufacturing: State-of-the-art and perspectives. *Addit. Manuf.* 2020, 36, 101538. [CrossRef]
- Meng, L.; McWilliams, B.; Jarosinski, W.; Park, H.Y.; Jung, Y.G.; Lee, J.; Zhang, J. Machine learning in additive manufacturing: A review. J. Miner. Met. Mater. Soc. 2020, 72, 2363–2377. [CrossRef]
- 110. Du, S.; Liu, C.; Huang, D. A shearlet-based separation method of 3D engineering surface using high definition metrology. *Precis. Eng.* **2015**, *40*, 55–73. [CrossRef]
- Syamsuddin. Formation and improvement of surface waviness for additive manufacturing 5A06 aluminium alloy component with GTAW system. *Rapid Prototyp. J.* 2019, 32, 91–100.
- 112. Golhin, A.P.; Tonello, R.; Frisvad, J.R.; Grammatikos, S.; Strandlie, A. Surface Roughness of As-Printed Polymers: A Comprehensive Review. *Int. J. Adv. Manuf.* 2023, 127, 987–1043. [CrossRef]
- Wi, K.; Suresh, V.; Wang, K.; Li, B.; Qin, H. Quantifying quality of 3D printed clay objects using a 3D structured light scanning system. *Addit. Manuf.* 2020, 32, 100987. [CrossRef]
- 114. Triantaphyllou, A.; Giusca, C.L.; Macaulay, G.D.; Roerig, F.; Hoebel, M.; Leach, R.K.; Tomita, B.; Milne, K.A. Surface texture measurement for additive manufacturing. *Surf. Topogr. Metrol. Prop.* **2015**, *3*, 024002. [CrossRef]
- 115. Suh, C.H.; Jung, Y.C.; Kim, Y.S. Effects of thickness and surface roughness on mechanical properties of aluminum sheets. *J. Mech. Sci. Technol.* 2010, 24, 2091–2098. [CrossRef]
- ISO 21920-2:2021; Geometrical Product Specifications (GPS)—Surface Texture: Profile—Part 2: Terms, Definitions and Surface Texture Parameters. ISO: Geneva, Switzerland, 2021. Available online: https://www.iso.org/standard/72226.html (accessed on 20 August 2023).
- ISO 25178-2:2021; Geometrical Product Specifications (GPS)—Surface Texture: Areal—Part 2: Terms, Definitions and Surface Texture Parameters. ISO: Geneva, Switzerland, 2021. Available online: https://www.iso.org/standard/74591.html (accessed on 20 August 2023).
- 118. Lee, S.H. Optimization of cold metal transfer-based wire arc process regression. Metals 2020, 10, 461. [CrossRef]
- 119. Li, Y.; Linke, B.S.; Voet, H.; Falk, B.; Schmitt, R.; Lam, M. Cost, sustainability and surface roughness quality—A comprehensive analysis of products made with personal 3D printers. *CIRP J. Manuf. Sci. Technol.* **2017**, *16*, 1–11. [CrossRef]
- 120. Magnien, J.; Cosemans, P.; Nutal, N.; Kairet, T. Current surface issues in additive manufacturing. *Plasma Process. Polym.* **2020**, 17, 1900154. [CrossRef]
- 121. Francesco, A.; Rob, S.; Marc, G.K.; Schmid, M.; Wegener, K. Selective Laser Sintering and Multi Jet Fusion: Process-induced modification of the raw materials and analyses of parts performance. *Addit. Manuf.* **2019**, *27*, 32–41. [CrossRef]
- Wu, D.; Wei, Y.; Terpenny, J. Surface roughness prediction in additive manufacturing using machine learning. In Proceedings of the ASME 2018 13th International Manufacturing Science and Engineering Conference, MSEC 2018, College Station, TX, USA, 18–22 June 2018; Volume 3, pp. 1–6. [CrossRef]
- 123. Silbernagel, C.; Aremu, A.; Ashcroft, I. Using machine learning to aid in the parameter optimisation process for metal-based additive manufacturing. *Rapid Prototyp. J.* 2020, *26*, 625–637. [CrossRef]
- Sendino, S.; Martinez, S.; Lartategui, F.; Gardon, M.; Lamikiz, A.; Gonzalez, J.J. Effect of powder particle size distribution on the surface finish of components manufactured by laser powder bed fusion. *Int. J. Adv. Manuf. Technol.* 2023, 124, 789–799. [CrossRef]
- 125. Delfs, P.; Tows, M.; Schmid, H.J. Optimized build orientation of additive manufactured parts for improved surface quality and build time. *Addit. Manuf.* 2016, 12, 314–320. [CrossRef]

- 126. Galati, M.; Rizza, G.; Defanti, S.; Denti, L. Surface roughness prediction model for Electron Beam Melting (EBM) processing Ti6Al4V. *Precis. Eng.* **2021**, *69*, 19–28. [CrossRef]
- 127. Barari, A.; Kishawy, H.A.; Kaji, F.; Elbestawi, M.A. On the surface quality of additive manufactured parts. *Int. J. Adv. Manuf. Technol.* **2017**, *89*, 1969–1974. [CrossRef]
- 128. Campbell, C.; Ying, Y. Learning with Support Vector Machines; Springer International Publishing: Berlin/Heidelberg, Germany, 2011.
- 129. Singh, T.; Kumar, P.; Misra, J.P. Surface roughness prediction modelling for wedm of aa6063 using support vector machine technique. *Mater. Sci. Forum* 2019, *969*, 607–612.
- Joshi, M.S.; Flood, A.; Sparks, T.; Liou, F.W. Applications of supervised machine learning algorithms in additive manufacturing: A review. In Proceedings of the 2019 Annual International Solid Freeform Fabrication Symposium, Austin, TX, USA, 12–14 August 2019; pp. 213–224.
- Mishra, A.; Jatti, V.S.; Sefene, E.M.; Paliwal, S. Explainable Artificial Intelligence (XAI) and Supervised Machine Learning-based Algorithms for Prediction of Surface Roughness of Additively Manufactured Polylactic Acid (PLA) Specimens. *Appl. Mech.* 2023, 4, 668–698. [CrossRef]
- 132. Jin, Z.; Shang, J.; Zhu, Q.; Ling, C.; Xie, W.; Qiang, B. RFRSF: Employee Turnover Prediction Based on Random Forests and Survival Analysis. In Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics); Springer International Publishing: Berlin/Heidelberg, Germany, 2020; Volume 12343, pp. 503–515. [CrossRef]
- Taunk, K.; De, S.; Verma, S.; Swetapadma, A. A brief review of nearest neighbor algorithm for learning and classification. In Proceedings of the 2019 International Conference on Intelligent Computing and Control Systems, ICCS 2019, Madurai, India, 15–17 May 2019; pp. 1255–1260.
- 134. Huang, M.; Jin, S.; Tang, Z.; Chen, Y.; Qin, Y. A Method for Predicting Surface Finish of Polylactic Acid Parts Printed Using Fused Deposition Modeling. *Processes* 2023, *11*, 1820. [CrossRef]
- 135. Dastres, R.; Soori, M. Artificial Neural Network Systems. *Int. J. Imaging Robot.* **2021**, 2021, 13–25. Available online: https://hal.science/hal-03349542 (accessed on 10 August 2023).
- Soler, D.; Telleria, M.; García-Blanco, M.B.; Espinosa, E.; Cuesta, M.; Arrazola, P.J. Prediction of Surface Roughness of SLM Built Parts after Finishing Processes Using an Artificial Neural Network. J. Manuf. Mater. Process. 2022, 6, 82. [CrossRef]
- 137. Boschetto, A.; Giordano, V.; Veniali, F. Surface roughness prediction in fused deposition modelling by neural networks. *Int. J. Adv. Manuf. Technol.* **2013**, *67*, 2727–2742. [CrossRef]
- 138. Wafa, F.; Abdulshahed, A.M. Prediction of surface roughness in additive manufacturing using artificial neural networks. In Proceedings of the International Engineering Conference (IEC-MU22), Misurata, Libya, 6 October 2022.
- Lakshmi, K.S.; Arumaikkannu, G. Evaluation of surface roughness in additive manufactured customised implant using artificial neural network based on 2D fourier transform—A machine vision approach. *Biomed. Res.* 2015, 26, S34–S40.
- Vahabli, E.; Rahmati, S. Improvement of FDM parts' surface quality using optimized neural networks—Medical case studies. *Rapid Prototyp. J.* 2017, 23, 825–842. [CrossRef]
- Mahapatra, S.S.; Sood, A.K. Bayesian regularization-based Levenberg-Marquardt neural model combined with BFOA for improving surface finish of FDM processed part. Int. J. Adv. Manuf. Technol. 2012, 60, 1223–1235. [CrossRef]
- 142. Vahabli, E.; Rahmati, S. Application of an RBF neural network for FDM parts' surface roughness prediction for enhancing surface quality. *Int. J. Precis. Eng. Manuf.* 2016, 17, 1589–1603. [CrossRef]
- 143. Jiang, J.; Hu, G.; Li, X.; Xu, X.; Zheng, P.; Stringer, J. Analysis and prediction of printable bridge length in fused deposition modelling based on back propagation neural network. *Virtual Phys. Prototyp.* **2019**, *14*, 253–266. [CrossRef]
- 144. Barrios, J.M.; Romero, P.E. Decision tree methods for predicting surface roughness in fused deposition modeling parts. *Materials* **2019**, *12*, 2574. [CrossRef]
- García Plaza, E.; López, P.J.N.; Torija, M.A.C.; Muñoz, J.M.C. Analysis of PLA geometric properties processed by FFF additive manufacturing: Effects of process parameters and plate-extruder precision motion. *Polymers* 2019, 11, 1581. [CrossRef] [PubMed]
- 146. Pfleging, W. Multiobjective optimization of laser polishing of additively manufactured ti-6al-4v parts for minimum surface roughness and heat-affected zone. *Materials* **2022**, *15*, 3323. [CrossRef]
- 147. Kandananond, K. Optimization of fused filament fabrication system by response surface method. *Int. J. Metrol. Qual. Eng.* 2019, *11*, 4. [CrossRef]
- 148. Ulkir, O.; Akgun, G. Predicting and optimising the surface roughness of additive manufactured parts using an artificial neural network model and genetic algorithm. *Sci. Technol. Weld. Join.* **2023**, *28*, 548–557. [CrossRef]
- 149. Chen, H.; Yaseer, A.; Zhang, Y. Top Surface Roughness Modeling for Robotic Wire Arc Additive Manufacturing. *J. Manuf. Mater. Process.* **2022**, *6*, 39. [CrossRef]
- 150. Roberts, D.A.; Yaida, S.; Hanin, B. The Principles of Deep Learning Theory; Cambridge University Press: Cambridge, UK, 2022.
- 151. Janiesch, C.; Zschech, P.; Heinrich, K. Machine learning and deep learning. *Electron. Mark.* 2021, 31, 685–695. [CrossRef]
- 152. Saxena, A. An Introduction to Convolutional Neural Networks. Int. J. Res. Appl. Sci. Eng. Technol. 2022, 10, 943–947. [CrossRef]
- 153. Abhilash, P.M.; Ahmed, A. Convolutional neural network–based classification for improving the surface quality of metal additive manufactured components. *Int. J. Adv. Manuf. Technol.* **2023**, *126*, 3873–3885. [CrossRef]
- 154. Garland, A.P.; White, B.C.; Jared, B.H.; Heiden, M.; Donahue, E.; Boyce, B.L. Deep Convolutional Neural Networks as a Rapid Screening Tool for Complex Additively Manufactured Structures. *Addit. Manuf.* **2020**, *35*, 101217. [CrossRef]

- 155. Banadaki, Y.; Razaviarab, N.; Fekrmandi, H.; Sharifi, S. Toward Enabling a Reliable Quality Monitoring System for Additive Manufacturing Process using Deep Convolutional Neural Networks. *arXiv* **2020**, arXiv:2003.08749.
- 156. Williams, G.; Meisel, N.A.; Simpson, T.W.; McComb, C. Design repository effectiveness for 3D convolutional neural networks: Application to additive manufacturing. *J. Mech. Des. Trans. ASME* **2019**, *141*, 111701. [CrossRef]
- 157. Jiang, J.; Xiong, Y.; Zhang, Z.; Rosen, D.W. Machine learning integrated design for additive manufacturing. *J. Intell. Manuf.* 2022, 33, 1073–1086. [CrossRef]
- 158. Montavon, G.; Samek, W.; Müller, K.R. Methods for interpreting and understanding deep neural networks. *Digit. Signal Process. A Rev. J.* **2018**, 73, 1–15. [CrossRef]
- 159. Samek, W.; Binder, A.; Montavon, G.; Lapuschkin, S.; Müller, K.R. Evaluating the visualization of what a deep neural network has learned. *IEEE Trans. Neural Netw. Learn. Syst.* 2017, 28, 2660–2673. [CrossRef] [PubMed]
- Muhammad, W.; Kang, J.; Ibragimova, O.; Inal, K. Experimental investigation and development of a deep learning framework to predict process-induced surface roughness in additively manufactured aluminum alloys. *Weld. World* 2023, 67, 897–921. [CrossRef]
- Kwon, O.; Kim, H.G.; Ham, M.J.; Kim, W.; Kim, G.H.; Cho, J.H.; Kim, N.I.; Kim, K. A deep neural network for classification of melt-pool images in metal additive manufacturing. *J. Intell. Manuf.* 2020, *31*, 375–386. [CrossRef]
- Maleki, E.; Bagherifard, S.; Guagliano, M. Correlation of residual stress, hardness and surface roughness with crack initiation and fatigue strength of surface treated additive manufactured AlSi10Mg: Experimental and machine learning approaches. *J. Mater. Res. Technol.* 2023, 24, 3265–3283. [CrossRef]
- Goodfellow, I.; Pouget-Abadie, J.; Mirza, M.; Xu, B.; Warde-Farley, D.; Ozair, S.; Courville, A.; Bengio, Y. Generative adversarial networks. *Commun. ACM* 2020, 63, 139–144. [CrossRef]
- Liu, W.; Wang, Z.; Tian, L.; Lauria, S.; Liu, X. Melt pool segmentation for additive manufacturing: A generative adversarial network approach. *Comput. Electr. Eng.* 2021, 92, 107183. [CrossRef]
- 165. Ogoke, O.F.; Johnson, K.; Glinsky, M.; Laursen, C.; Kramer, S.; Barati Farimani, A. Deep-learned generators of porosity distributions produced during metal Additive Manufacturing. *Addit. Manuf.* **2022**, *60*, 103250. [CrossRef]
- 166. Zhang, Z.; Sahu, C.K.; Singh, S.K.; Rai, R.; Yang, Z.; Lu, Y. Machine learning based prediction of melt pool morphology in a laser-based powder bed fusion additive manufacturing process. *Int. J. Prod. Res.* **2023**, 2023, 3–17. [CrossRef]
- 167. Kim, T.; Kim, J.G.; Park, S.; Kim, H.S.; Kim, N.; Ha, H.; Choi, S.K.; Tucker, C.; Sung, H.; Jung, I.D. Virtual surface morphology generation of Ti-6Al-4V directed energy deposition via conditional generative adversarial network. *Virtual Phys. Prototyp.* 2023, 18, 2–18. [CrossRef]
- Chung, J.; Shen, B.; Kong, Z.J. Anomaly detection in additive manufacturing processes using supervised classification with imbalanced sensor data based on generative adversarial network. *J. Intell. Manuf.* 2023, 2022, 1–31. [CrossRef]
- 169. Michelucci, U. An Introduction to Autoencoders. arXiv 2022, arXiv:2201.03898.
- 170. Shi, Z.; Mandal, S.; Harimkar, S.; Liu, C. Surface Morphology Analysis Using Convolutional Autoencoder in Additive Manufacturing with Laser Engineered Net Shaping. *Procedia Manuf.* **2021**, *53*, 16–23. [CrossRef]
- 171. Plaat, A. Deep Reinforcement Learning; Springer Nature: Singapore, 2022.
- 172. Ogoke, F.; Farimani, A.B. Thermal control of laser powder bed fusion using deep reinforcement learning. *Addit. Manuf.* **2021**, 46, 102033. [CrossRef]
- 173. Xia, C.; Pan, Z.; Polden, J.; Li, H.; Xu, Y.; Chen, S. Modelling and prediction of surface roughness in wire arc additive manufacturing using machine learning. *J. Intell. Manuf.* 2022, *33*, 1467–1482. [CrossRef]
- 174. Kandananond, K. Surface roughness prediction of FFF-fabricated workpieces by artificial neural network and Box-Behnken method. *Int. J. Metrol. Qual. Eng.* 2021, 12, 17. [CrossRef]
- 175. Koo, J.; Lee, S.; Baek, A.M.C.; Park, E.; Kim, N. Downskin Surface Roughness Prediction with Machine Learning for As-Built CM247LC Fabricated Via Powder Bed Fusion with a Laser Beam. *3D Print. Addit. Manuf.* **2023**. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.